Corporate philanthropy, public awareness, and the cost of equity capital: Evidence from China

Jun Xie and Junyi Chen*

This study examines whether a firm's philanthropic behavior affects its cost of equity capital and whether public awareness about the firm influences this effect in the context of China. We find that firms experience an increase in their cost of equity capital when the probability or the amount of philanthropic giving increases, and this effect is moderated when firms have high public awareness. Furthermore, for firms in the eastern region or those in less competitive industries, the positive effect of corporate philanthropic giving on the cost of equity capital can be moderated as the firms' public awareness increases.

 $\it Key Words\colon$ Corporate Philanthropy; Cost of Equity Capital; Public Awareness; Corporate Social Responsibility.

JEL Classification Numbers: G30, P31, M14.

1. INTRODUCTION

Corporate social responsibility (CSR) has become an important criterion for business practices. As a unique component of CSR activities, corporate philanthropic giving (CPG) continues its growth momentum worldwide. Firms tend to engage in philanthropic activities for both altruistic and forprofit motivations (Sánchez, 2000; Maas and Liket, 2011; Hogarth et al., 2018; Yang and Tang, 2020), and the extant literature has given particular attention to the economic consequences of corporate philanthropy (Brammer and Millington, 2005; Godfrey, 2005; Wang and Qian, 2011; Chen et al., 2018b). However, whether CPG is value-increasing, decreasing, or

^{*} Xie: Corresponding author. Associate Professor, Department of Finance, School of Economics and Management, Wuhan University, Hubei, 430072, China. E-mail: syakun1979@whu.edu.cn. Acknowledgements: We are grateful to the editor and an anonymous reviewer of the journal for their valuable comments that have helped us to improve our paper. We acknowledge the financial support from the National Natural Science Foundation of China (No. 71661137003) and the Fundamental Research Funds for the Central Universities. Chen: Department of Finance, School of Economics and Management, Wuhan University, Hubei, 430072, China.

neutral still remains controversial. On the one hand, previous studies suggest that CPG promotes a firm's public image, strengthens its relationship with stakeholders, and thus, enhances its future performance (Berman et al., 1999; Godfrey, 2005; Lev et al., 2010; Chen et al., 2018b). On the other hand, empirical results also show the possibility that CPG is conducted by self-interested managers to extract wealth from shareholders, promote individual reputation, and advance their personal careers, which may worsen the firms' financial performance (Wang and Coffey, 1992; Marquis and Lee, 2013; Masulis and Reza, 2015; Wang et al., 2015).

As the expected rate of return on equity investment, the cost of equity capital is one of the most important factors which are closely related to firm value. Even though there is extensive literature investigating the financial outcomes of CPG from various perspectives, very few of them examine the capital market participants' perceptions of CPG and the effect of CPG on the firm's cost of equity capital is also subject to debate (Judd and Lusch, 2018; Zolotoy et al., 2019). Because CPG can be used as a tool by managers to pursue their own interests and as a result, the stockholders' value is expropriated, CPG will increase the cost of equity capital (Judd and Lusch, 2018). However, if CPG acts as a mechanism for developing a firm's relationship with stakeholders and thereby enhances investors' perceptions of a firm's characteristics, then the cost of equity capital would decrease because of the reduction of the firm's business risk (Zolotoy et al., 2019). Hence, the effect of CPG on the cost of equity capital is unclear and it is necessary to investigate the factors that may have significant impacts on the CPG's effect on the cost of equity capital. In this paper, we argue that the impact of CPG on the cost of equity capital depends on the capacity of such CPG activities to influence the firm's stakeholders, and the precondition for CPG to modify stakeholders' behavior is the stakeholders' awareness of the firm's CPG activities. Servaes and Tamayo (2013) advocate that the consumers are more likely to respond to a firm's CSR activities if they are aware of them. However, Pomering and Dolnicar (2009) indicate that consumers' awareness level of CSR is quite low so that a firm's CSR activities fail to prove their effectiveness in the marketplace. This phenomenon arouses our interests to re-examine the effects of CPG on the cost of equity capital by focusing on one indirect factor, the public awareness of the firm, and this study tries to find out how public awareness adjusts the effects of CPG on the cost of equity capital.

Although some literature has examined the impact of CSR on the cost of equity capital (Dhaliwal et al., 2011; Xu et al., 2015), we argue that it is necessary to re-examine this issue by focusing on CPG. First, CSR is a broad concept which includes various aspects of social responsibilities that firms are expected to fulfill. Thus, the components within CSR may have different and sometimes even competing influential mechanisms. Sec-

ond, CSR measures consist of many subjective factors and their quality always raise questions. Hence, it is significant to separate the individual components of CSR (Moser and Martin, 2012; Judd and Lusch, 2018).

In this study, we argue that public awareness, which is proxied by advertising intensity, can have significant impacts on the effect of CPG on the cost of equity capital. The public awareness of a firm is associated with visibility, familiarity, and social judgment. The psychological literature shows that human behaviors and cognitive activities are largely influenced by the perceivers' familiarity with or awareness of the target (Fiske and Cox, 1979; Reis et al., 2011; Finkel et al., 2015). A higher level of brand familiarity leads to greater brand trust, and consumers show an overwhelming preference and purchase intention for the brands that they are most aware of (Keller, 1993; Laroche et al., 1996; Macdonald and Sharp, 2000; Ha and Perks, 2005; Esch et al., 2006; Foroudi et al., 2018). Accordingly, firms tend to invest in advertising activities to reduce the information gap between the public and themselves, and improve public awareness (Servaes and Tamayo, 2013). In finance, the extant literature argues that investment decision-making is highly correlated with investors' awareness of the stock. Grullon et al. (2004) provide evidence that firms with higher advertising intensity attract a larger number of both individual and institutional investors. Yung and Nafar (2017) and Meng et al. (2020) demonstrate that the expected returns of securities are negatively associated with the level of investors' awareness because of the compensation for being imperfectly diversified. The home bias theory also proposes that investors prefer to hold the stocks of locally headquartered firms, and that they ignore foreign investment opportunities (French and Poterba, 1991; Coval and Moskowitz, 1999; Nieuwerburgh and Veldkamp, 2009; Levy and Levy, 2014). Riff and Yagil (2020) examine the effects of branding and location on the willingness to invest, and their results show that people are more willing to invest in local and high-branded firms.

The extant literature on the effect of CPG on the cost of equity capital is limited to western developed countries. As the largest emerging country, China is an especially interesting setting for this research topic because its economic and institutional environments are quite different from those of western countries. First, nearly 69% of charitable giving in China is attributable to corporate donors, which is much higher than the proportion of U.S. corporate donors. However, charitable donations from Chinese individuals' account for merely 11%, which is far lower than the average level worldwide (Schrader and Xie, 2016). Hence, in China, CPG plays a major role in the area of charitable giving. Second, the motives for CPG are much more complicated in China, where governments at all levels have significant influence on CPG decisions. Due to market inefficiencies, it is difficult for small and medium-sized enterprises and non-state-owned en-

terprises (NSOEs) to obtain economic resources via market competition. To reduce financing constraints or obtain better investment opportunities, firms have to build political connections with the government through charitable giving (Wang and Qian, 2011; Gao et al., 2012; Chen et al., 2015; Chen et al., 2018b). Furthermore, firms that receive government support are also expected to reciprocate by engaging in more CPG, such that politically connected firms are more likely to respond to the government's call for CPG (Li et al., 2015; Yang and Tang, 2020). Third, China provides a unique environment based on the co-existence of both state-owned enterprises (SOEs) and NSOEs (Allen et al., 2005). More than half of the listed companies in China are still state-owned and their CEOs are often appointed by the government. These firms are politically connected and need to achieve political and social targets aside from their operational objectives (Kato and Long, 2006; Chen et al., 2011b; Xu et al., 2015). Therefore, an investigation in the Chinese context can present different and remarkable results and improve our understanding of the role of CPG in economies in which politics interact widely and deeply with firms' behaviors.

We examine the effect of CPG on the cost of equity capital based on a sample of A-share-listed Chinese companies from 2004 to 2017. The empirical results demonstrate that firms experience an increase in their cost of equity capital when the probability or the amount of philanthropic giving increases. However, this positive effect is moderated when firms have high public awareness. These results are consistent with the idea that firms can benefit from CPG if their behaviors are highly visible. Second, we divide the sample into SOE and NSOE subsamples and find that the moderating effect of public awareness exists only for NSOEs. Furthermore, in cases where firms are located in the eastern region or are engaged in less competitive industries, the positive effect of corporate philanthropy on the cost of equity capital is moderated by an increase in public awareness. Our results are consistent with a battery of robustness tests.

Our research contributes to the literature in several ways. First, this study advances our understanding of the determinants of a firm's cost of equity capital. The extant literature tries to explain the variation of the cost of equity capital by focusing on internal control quality (Ashbaugh-Skaife et al., 2009; Gao and Jia, 2017; Khlif et al., 2019), ownership structure (Chu et al., 2014; Boubakri et al., 2016; Chu et al., 2019), information disclosure (Richardson and Welker, 2001; Dhaliwal et al., 2011; Li and Liu, 2018), and political connection (Boubakri et al., 2012; Pham, 2019). This study provides evidence that public awareness can be an important moderating factor affecting a firm's cost of equity capital. Second, this study enriches the philanthropy literature on the consequences of CPG activities. To the best of our knowledge, only two articles discuss how CPG affects a firm's cost of equity capital, and they draw divergent results in the U.S.

context (Judd and Lusch, 2018; Zolotov et al., 2019). By documenting the positive relationship between CPG and the cost of equity capital based on the data of Chinese listed companies, we show that CPG increases a firm's equity cost and provide another perspective to understand the passive consequences of philanthropy. Third, our research underlines the moderating effect of public awareness on the effect of CPG on the firm's cost of equity capital. We argue that investors' responses to CPG are subject to their familiarity with the firm and public awareness is an important factor that affects the influence of CPG on the cost of equity capital. Our results show that the positive effect of CPG on the cost of equity capital is mitigated as public awareness increases, which indicates that an "information-friendly" environment is necessary for firms to reduce business risks that originate from CPG involvement. Furthermore, we find that the moderating effect of public awareness is subject to other conditions such as ultimate ownership, the region in which the firm is located, and the conditions of market competition. Fourth, China is an especially interesting setting for this research because its economic and institutional environments are quite different from those in developed countries. Due to the immature development of market mechanisms and legal protection in China, CPG can be used as a tool to hide environmental misconduct (Du, 2015), obtain crucial resources (Chen et al., 2018b), or improve CEO's personal welfare (Masulis and Reza, 2015). Moreover, government intervention plays an important role in firms' CPG decision (Li et al., 2015). Hence, the incentives and economic outcomes of CPG are rather confusing in China, and this study provides unique insight on the role of CPG in emerging markets.

The remainder of this article is organized as follows. Section 2 discusses the prior literature and develops our hypothesis. Section 3 describes our data and methodology. Section 4 provides the empirical results and Section 5 presents the robustness tests. Finally, Section 6 concludes.

2. LITERATURE REVIEW AND HYPOTHESES DEVELOPMENT

The existing literature documents that CPG can be used by CEOs in pursuit of their personal goals rather than for stockholder value maximization, and the agency costs inherent in CPG can be a type of risk that increases firms' cost of equity capital (Masulis and Reza, 2015; Judd and Lusch, 2018). Philanthropic giving is often utilized by self-interested managers to promote their personal reputation, build social networks, and advance their careers (Werbel and Carter, 2002; Wang et al., 2008; Masulis and Reza, 2015). For example, Werbel and Carter (2002) examine the relationship between CEOs' personal interests and philanthropic giving and find that CEOs' membership in non-profit organizations increases a firm's donation

toward such groups. By donating a portion of the firm's profits to charitable causes, CEOs may gain access to elite circles or enhance their social power (Atkinson and Galaskiewicz, 1988; Marquis et al., 2007). In addition, narcissistic CEOs have stronger personal need to attract public attention and gain praise; thus, they are more likely to seek recognition through philanthropic behavior (Petrenko et al., 2016). Corporate donations can also be authorized by top managers out of an altruistic belief, and such philanthropic behavior indulges the manager's wish for "doing good" while leading to shareholders' loss (Brown et al., 2006). In the above-mentioned cases, the welfare of shareholders is violated because the firm's cash is spent to satisfy managers' personal preferences instead of rewarding shareholders or making investments in profitable projects. As such, these opportunistic CPG behaviors eventually lead to a decrease in firm value. Masulis and Reza (2015) find that the market valuation of firm cash holdings is lower if a firm engages in philanthropic activities, and this relation is more pronounced for firms with poor governance. Based on the data of Australian listed firms, Hogarth et al. (2018) indicate that a firm's Tobin's Q falls by approximately 0.413% for every cent spent on CPG. Wang et al. (2008) propose that CPG improves a firm's financial performance within certain limits; however, the benefits of CPG is offset by constantly growing costs as CPG increases beyond a certain level. Hence, the existence of agency conflicts and the exorbitant costs caused by CPG may augment the firm's business risk, and CPG would increase a firm's cost of equity capital.

However, previous literature also provides evidence that CPG can reduce a firm's business risk. Based on the strategic philanthropy theory, commitment to corporate philanthropy not only benefits the community, but also the firm itself (Varadarajan and Menon, 1988; Saiia et al., 2003; Liket and Maas, 2016; Chen et al., 2020). These studies argue that CPG helps a firm improve its reputation (Godfrey, 2005; Maas and Liket, 2011; Gardberg et al., 2019), enhance employee morale (Brown et al., 2006; Balakrishnan et al., 2011), maintain customer loyalty (Zhang et al., 2010; Mandhachitara and Poolthong, 2011; Aramburu and Pescador, 2019), and build political connections (Scherer and Palazzo, 2007; Wang and Qian, 2011; Chen et al., 2020). By engaging in CPG activities, a firm can generate moral capital and build its reputation as being honest and reliable (McWilliams and Siegel, 2001; Brammer and Millington, 2005; Zhang et al., 2010). Customers believe that the products of socially responsible companies are of higher quality; thus, they prefer to buy goods from companies that make more charitable donations (Lee and Shin, 2010). Accordingly, CPG activities are associated with subsequent sales growth (Lev et al., 2010). Furthermore, CPG motivates employees' efforts, boosts their loyalty, and increases the efficacy of implicit contracts (Brown et al., 2006; Balakrishnan et al., 2011). Moreover, socially responsible firms also have

incentives to protect their reputation and tend to behave well in subsequent business activities. As Chen et al. (2020) put forward, firms that engage in philanthropic activities have stronger willingness to prevent future corporate misconduct, and these firms are less likely to be penalized by the China Securities Regulatory Commission (CSRC). In addition, investors believe that socially responsible firms are less likely to shirk their responsibility to shareholders; thus, agency conflicts are less severe in these firms (Chen et al., 2018b; Zolotoy et al., 2019). Overall, considering the effects of CPG on improving a firm's relationship with stakeholders and reducing corporate misconduct, the business risk of a firm could be mitigated as it engages in CPG activities. Hence, CPG can lower a firm's cost of equity capital.

The motives of firms in making charitable donations are even more intricate in China, where the institutional environment is less developed and government intervention plays an important role in economy. Due to geographic or economic reasons, firms cannot control some of the resources necessary for their development. However, engaging in CPG can help firms build political connections to gain access to valuable resources that are crucial to their long-term growth (Wang and Qian, 2011; Chen et al., 2018b). Chen et al. (2015) find that CPG provides firms an advantage in gaining access to bank loans, and this relationship is stronger for NSOEs because SOEs enjoy privileges in the Chinese banking system and are less likely to curry favor with the government through philanthropic activities. Hence, the government protection and preferential benefits that firms obtain through philanthropic giving may decrease their business risk. Meanwhile, the Chinese government needs resources from society to realize its social responsibilities, and thus encourages firms to make charitable donations. Li et al. (2015) illustrate that politically connected firms have motives to reciprocate government support and that they are subject to stronger political intervention. Hence, politically connected firms tend to contribute more to philanthropic causes, which may add burden to the firms. In addition, Yang and Tang (2020) argue that entrepreneurs who hold a political office have stronger pro-social values and sense of social responsibility; thus, their firms are more likely to donate for the "common good" and ask for nothing in return. This altruistic, philanthropic behavior increases the firm's costs and business risk. Moreover, due to the loopholes and grey areas in China's institutional environment, CPG could be used to divert public attention and disguise corporate misconduct (Chen et al., 2018a). Du (2015) indicates that incidences of corporate environmental misconduct are positively connected to CPG, which implies that misconduct-dressing is also an incentive to engage in CPG. Thus, we propose the following hypotheses.

HYPOTHESIS 1a. Firms that engage in CPG activities have a higher cost of equity capital relative to firms that do not engage in CPG activities

in a given year, and the cost of equity capital increases as the amount of CPG increases.

HYPOTHESIS 1b. Firms that engage in CPG activities have a lower cost of equity capital relative to firms that do not engage in CPG activities in a given year, and the cost of equity capital decreases as the amount of CPG increases.

Next, we investigate the influence of public awareness on the impact of CPG on a firm's cost of equity capital. Previous literature on social psychology has highlighted the importance of available information in shaping people's perception of others (Fiske and Cox, 1979). The perceiver relies on available cognitive resources, such as observed actions, to make inferences about a person's internal characteristics (Jones and Davis, 1965). However, cognitive resources are sometimes insufficient, and thus hinder the perceivers' capacity to develop an accurate understanding of an item that receives limited attention (Falkinger, 2008). Following this logic, the prerequisite for CSR to obtain feedback is that such activities must first be noticed; however, consumers' awareness of CSR is quite low, such that a firm's CSR activities fail to prove their effectiveness in the marketplace (Pomering and Dolnicar, 2009). Schuler and Cording (2006) indicate that information intensity is a major factor that affects customers' awareness of CSR initiatives and further influences customers' brand attitudes. The extant literature suggests that advertising can enhance a firm's information environment and raise the awareness of stakeholders who are interested in a firm's CSR attributes (McWilliams and Siegel, 2001; Servaes and Tamayo, 2013). Based on the data of U.S. firms, Servaes and Tamayo (2013) find that firm value is positively associated with CSR for firms with high customer awareness, whereas the relation turns either negative or insignificant for firms with low customer awareness. Rhou et al. (2016) indicate that the positive relation between CSR initiatives and financial performance is more pronounced when public awareness is high. We reason that a high level of public awareness strengthens the stakeholders' capacity to respond to a firm's CPG activities and enables CPG to produce economic benefits. Hence, the business risk arising from CPG activities is mitigated as public awareness increases. In addition, firms with high public awareness attract more institutional investors and have higher liquidity of their common stock, which can improve corporate governance and reduce misconduct (Grullon et al., 2004; Mizuno, 2010; Brogaard et al., 2017). High public awareness of a firm creates better visibility for the firm, and firms with high visibility face greater pressure from public scrutiny; thus, they are more likely to behave well, thereby minimizing business risk (Gupta et al., 2018).

Based the discussion above, we predict that the business risk arising from CPG activities is lower for firms with higher public awareness. Therefore,

we infer that the interaction term between CPG and public awareness is negatively associated with a firm's cost of equity capital. We state our hypotheses 2a and 2b as follows:

HYPOTHESIS 2a. If CPG activities positively affect a firm's cost of equity capital, then the effect would be mitigated as the level of public awareness increases.

HYPOTHESIS 2b. If CPG activities negatively affect a firm's cost of equity capital, then the effect would be enhanced as the level of public awareness increases.

3. RESEARCH DESIGN

3.1. Sample selection and data sources

This study employs panel data of Chinese firms listed on the Shanghai and Shenzhen Stock Exchanges for period 2004-2017. We collect our data on CPG from the China Stock Market and Accounting Research (CS-MAR) Database. Financial data and industry affiliation data are also taken from the CSMAR database, and the consensus earnings forecasts data is provided by the Wind Financial Database. The CPG and other financial data are collected from the annual reports of listed companies by the CSMAR database. The CSMAR database makes use of the industry classification scheme prescribed by the China Securities Regulatory Commission (CSRC). The Wind Financial Database has a wide coverage of analysts' earnings forecasts of listed firms based on analyst reports and the consensus earnings forecasts are calculated as the average of the analysts' earnings forecasts. Following Chen et al. (2011b) and Xie (2015), a firm is considered as an SOE if it is ultimately controlled by the Chinese central government; local governments at the provincial, municipal, and county level; or other governmental institutions. Conversely, a firm is considered as a NSOE if it is ultimately controlled by an individual or a non-state entity (e.g., town-village enterprise, foreign enterprise, or other non-state-controlled enterprise). The ultimate controlling shareholder data is obtained from the CSMAR database.

We exclude financial firms and special treatment (ST) firms because both of them are subject to special regulations and ST firms suffer from financial and operational problems. We use the mean of four different estimates of the cost of equity capital to mitigate measurement errors; thus, some observations with insufficient data to estimate the cost of equity capital are eliminated. All continuous variables are winsorized at the top and bottom 5%. This procedure yields a final sample of 12,612 firm-year observations, which accounts for 36.6% of the 34,466 firm-year observations in the initial data (excluding financial firms). We use unbalanced panel data in our analysis, because the number of listed firms in China has been

 ${\bf TABLE~1.}$ The yearly and industrial distributions of the sample.

	N	The proportion of total sample
Panel A: Distribution		I I
2004	188	1.49%
2005	247	1.96%
2006	384	3.04%
2007	422	3.35%
2008	543	4.31%
2009	767	6.08%
2010	1042	8.26%
2011	1203	9.54%
2012	1165	9.24%
2013	1087	8.62%
2014	1187	9.41%
2015	1318	10.45%
2016	1447	11.47%
2017	1612	12.78%
Total	12612	100.00%
Panel B: Distribution	by industry	
A	201	1.59%
В	379	3.01%
C	8236	65.30%
D	376	2.98%
E	389	3.08%
F	685	5.43%
G	345	2.74%
H	43	0.34%
I	707	5.61%
K	555	4.40%
L	176	1.40%
M	83	0.66%
N	127	1.01%
O	10	0.08%
P	7	0.06%
Q	32	0.25%
R	161	1.28%
S	100	0.79%
Total	12612	100.00%

Notes: This table presents the yearly and industrial distributions of the sample. Industry categories are prescribed by the CSRC, including agriculture, forestry, livestock rearing, and fishing (A); mining (B); manufacturing (C); electric power, gas, and water production and supply (D); construction (E); wholesale and retail (F); transportation, storage and postal services (G); hotel and catering services (H); information transmission, software and information technology services (I); finance (J); real estate (K); leasing and commercial services (L); scientific research and technical services (M); water conservancy, environment and public facilities management (N); residential service, repairing and other services (O); education (P); health and social work (Q); culture, sports and entertainment (R); and conglomerate (S).

increasing rapidly in the past years. Table 1 shows the yearly and industrial distributions of our sample. The number of firms demonstrates a nearly monotonically increasing trend from 2004 to 2017, and the manufacturing industry represents a strong majority.

3.2. Variable construction

3.2.1. Measures of CPG

Consistent with the related literature (e.g., Wang and Qian, 2011; Zhang et al., 2016; Chen et al., 2018b), we employ two measures for CPG in our main empirical tests. The first measure PHI_D is a dummy variable that equals 1 if the amount of a firm's donation expenditures in a specific fiscal year is positive and 0 otherwise. To alleviate skewness, we define our second measure, PHI_M , as a continuous variable calculated as the natural logarithm of 1 plus the amount of a firm's donation expenditures in a specific fiscal year. PHI_D measures whether a firm engages in any CPG activities in a fiscal year, while PHI_M presents the amount of CPG of a firm in a fiscal year.

3.2.2. Measures of the cost of equity capital

Following Chen et al. (2011a), Xu et al. (2015), and Dhaliwal et al. (2016), we construct our dependent variable using the ex-ante implied cost of equity capital which is defined as the internal rate of return that equates the present value of the expected future cash flows to the current market price. Compared with ex-ante measures, the ex-post realized returns are noisier and face more severe estimation errors because they also capture shocks to a firm's growth opportunities (Stulz, 1999). This defect is magnified in Chinese stock market where many listed firms are at the growth stage (Kim et al., 2015). Hence, the ex-ante measures that control for both future cash flow and growth potential (Hail and Leuz, 2006) are better proxies for the cost of equity capital in the Chinese stock market. We use four different measures which are constructed by Claus and Thomas $(2001) (R_{CT})$, Gebhardt et al. $(2001) (R_{GLS})$, Easton $(2004) (R_{PEG})$, and Ohlson and Juettner-Nauroth (2005) (R_{OJN}) . The four models can avoid the long time-series requirement of realized returns to estimate an unbiased expected return, because they use the analysts' future earnings forecasts to estimate the cost of equity capital. The first two models are based on the residual income valuation model in Ohlson (1995), while the latter two models are developed from Ohlson and Juettner-Nauroth (2000)'s abnormal earnings growth valuation model. All of these models are essentially variations of discounted cash flow valuation. However, they differ in terms

of the duration of forecast horizons, assumptions about the future earnings growth rate, and exploitation of analyst forecasts. To address the concern of spurious results caused by one particular model, we use the arithmetic average of the four commonly accepted models to calculate the cost of equity capital (Hail and Leuz, 2006; Chen et al., 2011a; Dhaliwal et al., 2016; Judd and Lusch, 2018).

The variables used in the following four models are as follows. P_t is the market price of a firm's stock at the end of year t, while B_t is the book value per share of a firm at the beginning of year t. $FEPS_{t+i}$ is the forecast earnings per share of a firm in year t+i. POUT is the forecast dividends payout ratio of a firm and is calculated as a firm's historical three-year average dividend payout ratio. R_f is the yield on a 10-year Treasury note in June of year t. g is calculated as R_f minus 3%.

 R_{GLS} is estimated from the model of Gebhardt et al. (2001).

$$P_{t} = B_{t} + \sum_{i=1}^{11} \frac{(FROE_{t+i} - R_{GLS}) \times B_{t+i-1}}{(1 + R_{GLS})^{i}} + \frac{(FROE_{t+12} - R_{GLS}) \times B_{t+11}}{R_{GLS} \times (1 + R_{GLS})^{11}}$$

$$\tag{1}$$

where $B_{t+i} = B_{t+i-1} + (1 - POUT) \times FEPS_{t+i}$. For the first three years, $FROE_{t+i}$ is equal to $FEPS_{t+i}/B_{t+i-1}$. Thereafter, we forecast $FROE_{t+i}$ using a linear interpolation to the industry median ROE from the 4^{th} year to the 12^{th} year. From the 12^{th} year onward, $FROE_{t+i}$ is assumed to be constant.

 R_{CT} is estimated from the model of Claus and Thomas (2001).

$$P_{t} = B_{t} + \sum_{i=1}^{5} \frac{FEPS_{t+i} - R_{CT} \times B_{t+i-1}}{(1 + R_{CT})^{i}} + \frac{(FEPS_{t+5} - R_{CT} \times B_{t+4}) \times (1 + g)}{(R_{CT} - g) \times (1 + R_{CT})^{5}}$$
(2)

where $B_{t+i} = B_{t+i-1} + (1 - POUT) \times FEPS_{t+i}$. $FEPS_{t+4}$ and $FEPS_{t+5}$ are calculated based on $FEPS_{t+3}$ and the long-term earnings growth rate. The long-term earnings growth rate is calculated as the implied earnings growth rate from $FEPS_{t+1}$ to $FEPS_{t+3}$.

 R_{OJN} is estimated from the model of Ohlson and Juettner-Nauroth (2005).

$$P_{t} = \frac{FEPS_{t+1}}{R_{OJN}} + \frac{(FEPS_{t+2} - FEPS_{t+1} - R_{OJN} \times FEPS_{t+1} \times (1 - POUT))}{R_{OJN} \times (R_{OJN} - g)}$$
(3)

which suggests that

$$R_{OJN} = A + \sqrt{A^2 + \frac{FEPS_{t+1}}{P_t} \times \left(\frac{FEPS_{t+2} - FEPS_{t+1}}{FEPS_{t+1}} - g\right)}$$
 (4)

where $A = \frac{1}{2} \times \left(g + \frac{FEPS_{t+1} \times POUT}{P_t}\right)$. The calculation of this model requires that $FEPS_{t+1} > 0$ and $FEPS_{t+2} > 0$.

 R_{PEG} is estimated from the model of Easton (2004).

$$P_{t} = \frac{FEPS_{t+2} - FEPS_{t+1} + R_{PEG} \times FEPS_{t+1} \times POUT}{R_{PEG}^{2}}$$
 (5)

The implementation of this model requires that $FEPS_{t+2} \ge FEPS_{t+1} > 0$.

3.2.3. Measures of public awareness

Following Wang and Qian (2011) and Servaes and Tamayo (2013), we use the intensity of a firm's advertising as a proxy for its public awareness. Intensive advertising improves a firm's visibility and attracts more attention from the public; thus, it increases the likelihood of stakeholders noticing the firm's CPG activities (Wang and Qian, 2011). Following Zhang et al. (2010) and Kashmiri et al. (2019), we define ADVERT as the ratio of advertising expenditures to total asset. The data on advertising expenditures is shown as a sub-item of sales expenses in annual reports, which is the sum of various advertising expenditures (e.g., TV, newspaper, billboards, and internet advertising costs). Similar to Servaes and Tamayo (2013), the data of advertising expenditures are missing for nearly half of the observations (47.99%) because firms do not need to disclose advertising expenditures if they are immaterial (Servaes and Tamayo 2013, p. 1051). Hence, we follow prior studies (e.g., Fee et al., 2009; Servaes and Tamayo, 2013) and set advertising expenditure to zero when the data are missing. We define HighAwareness as a dummy variable that equals 1 if the ADVERT of a firm is above the sample mean in a fiscal year and 0 otherwise.

3.2.4. Control variables

Following the related studies (Chen et al., 2011a; Xu et al., 2015; Dhaliwal et al., 2016; Judd and Lusch, 2018), we control for firm size, the bookto-market ratio, leverage, price momentum, liquidity, and forecast long term growth rate in our regressions.

Fama and French (1992) find that stock returns are negatively correlated with firm size and are positively correlated with book-to-market equity

TABLE 2.
Variable definitions.

	variable definitions.						
Variable	Definition						
Dependent Variab	les						
$\overline{R_{AVE}}$	The arithmetic average of the four individual estimates of the cost of						
	equity capital. (i.e., R_{GLS} , R_{CT} , R_{OJN} , and R_{PEG}).						
R_{GLS}	The cost of equity capital estimated from the model of Gebhardt et al.						
	(2001).						
R_{CT}	The cost of equity capital estimated from the model of Claus and						
	Thomas (2001).						
R_{OJN}	The cost of equity capital estimated from the model of Ohlson and						
	Juettner-Nauroth (2005).						
R_{PEG}	The cost of equity capital estimated from the model of Easton (2004).						
Independent Varia	ables						
$\overline{PHI_D}$	A dummy variable that equals 1 if the amount of a firm's donation						
	expenditures in a specific fiscal year is positive and 0 otherwise.						
PHI_M	A continuous variable calculated as the natural logarithm of one plus						
	the amount of a firm's donation expenditures in a specific fiscal year.						
ADVERT	Ratio of advertising expenditures to total asset of a firm.						
High Awareness	A dummy variable that equals one if the $ADVERT$ of a firm is above						
	the sample mean in a fiscal year and zero otherwise.						
Control Variables							
\overline{MVE}	Natural logarithm of the firm's market value of equity.						
BTM	Ratio of the book value of equity to the market value of equity.						
LEV	Book value of total liabilities deflated by the book value of total assets.						
MTUM	Stock return over the fiscal year.						
LIQUID	Ratio of the number of shares traded in a year to the total shares						
	outstanding at the end of that year.						
LTG	Difference between the mean of two-year-ahead analyst consensus EPS						
	forecast and the mean of one-year-ahead analyst consensus EPS forecast						
	divided by the mean of one-year-ahead analyst consensus EPS forecast.						

value. Firm size (MVE) is calculated as the natural logarithm of the market value of equity, while the book-to-market ratio (BTM) is measured as the ratio of the book value of equity to the market value of equity. Because Dhaliwal et al. (2016) show that the cost of equity capital is positively related to the firm's leverage, we also include leverage ratio (LEV), which is measured as the book value of total liabilities deflated by the book value of total assets. The estimates of ex-ante cost of equity capital may be biased because of the use of analysts' forecasts, which are sluggish with respect to the information in past returns on equity (Guay et al., 2011). Hence, we

include price momentum (MTUM) as a control variable, calculated as the stock return over the fiscal year. We also control for liquidity (LIQUID) which is the ratio of the number of shares traded in a year to the total shares outstanding at the end of that year, as prior studies suggest that expected returns should be higher for illiquid stock to compensate for higher transaction costs (Amihud and Mendelson, 1986). Following Chen et al. (2011a), we include forecast long-term growth rate (LTG) to control for the potential estimation biases of the ex-ante cost of equity capital and LTG is calculated as the difference between the mean of the two-year-ahead analyst consensus EPS forecast and the mean of the one-year-ahead analyst consensus EPS forecast. Finally, we control for industry and time fixed effects by introducing industry and year dummies.

The detailed definitions of variables are presented in Table 2.

TABLE 3.Descriptive statistics.

Variable	N	Mean	Median	Min	Max	Standard
						deviation
$\overline{R_{AVE}}$	12612	0.0867	0.0836	0.0430	0.1474	0.0282
R_{GLS}	12612	0.0463	0.0442	0.0197	0.0827	0.0174
R_{CT}	12612	0.0825	0.0771	0.0312	0.1636	0.0353
R_{PEG}	12612	0.1075	0.1046	0.0539	0.1767	0.0331
R_{OJN}	12612	0.1096	0.1067	0.0554	0.1793	0.0333
PHI_D	12612	0.9034	1.0000	0.0000	1.0000	0.2954
PHI_M	12612	11.4378	12.5179	0.0000	15.9277	4.1890
ADVERT	12612	0.0025	0.0000	0.0000	0.0205	0.0053
MVE	12612	15.7515	15.7030	14.2489	17.5110	0.8888
BTM	12612	0.6110	0.6153	0.2342	0.9724	0.2189
LEV	12612	0.4238	0.4226	0.0997	0.7636	0.1977
MTUM	12612	0.2300	0.0019	-0.4779	1.7598	0.5918
LIQUID	12612	3.1019	2.5795	0.6270	7.9124	2.0216
LTG	12612	0.4133	0.3135	0.0988	1.4059	0.3217

3.3. Summary statistics

Table 3 provides the descriptive statistics for the variables used in our main analysis. The mean of R_{AVE} is 8.67%, and the means of RPEG and R_{OJN} (10.75% and 10.96%, respectively) are higher than those of R_{GLS} and R_{CT} (4.63% and 8.25%, respectively), which are similar to the prior research (Hail and Leuz, 2006; Xu et al., 2015; Dhaliwal et al., 2016;

Judd and Lusch, 2018). The mean of the CPG measure PHI_D is 0.9034, suggesting that 90.34% of the firms in our sample are involved in CPG activities. On average, the firms in our sample have a market value of equity (in logarithmic form) of 15.7515, book-to-market ratio of 61.10%, leverage ratio of 42.38%, momentum ratio of 23.00%, liquidity ratio of 3.1019, and long-term growth rate of 41.33%, which are similar to the extant literature (Xu et al., 2015; Dhaliwal et al., 2016; Judd and Lusch, 2018).

TABLE 4.

				Corr	elations ma	trix.					
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)
(1) R_{AVE}	1.0000										
(2) $PHI_{-}D$	0.0614^{***}	1.0000									
(3) $PHI_{-}M$	0.1173^{***}	0.8928^{***}	1.0000								
(4) ADVERT	-0.0809^{***}	0.0468^{***}	0.0985^{***}	1.0000							
(5) SOE	0.0644^{***}	-0.0054	0.0175^{**}	-0.0772^{***}	1.0000						
(6) MVE	-0.1228***	0.0739***	0.2189***	0.1079^{***}	0.1349^{***}	1.0000					
(7) BTM	0.4740^{***}	0.0654***	0.1177^{***}	-0.1673^{***}	0.2318^{***}	-0.1758***	1.0000				
(8) LEV	0.2985^{***}	0.0611***	0.1312***	-0.1316***	0.3394***	0.1646^{***}	0.3990***	1.0000			
(9) MTUM	-0.2053^{***}	-0.0298***	-0.0447^{***}	-0.0668***	0.0457^{***}	0.1567^{***}	-0.3300***	0.0830***	1.0000		
(10) LIQUID	-0.1999^{***}	-0.0683^{***}	-0.1243***	-0.0688***	-0.0085	-0.1019^{***}	-0.3325***	0.0432***	0.3431***	1.0000	
(11) LTG	0.3588***	-0.0489^{***}	-0.0908***	-0.0996***	-0.0557***	-0.0929***	-0.0883***	0.1148***	0.1555***	0.1937***	1.0000

Notes: *, ** and *** represent significance at the 10%, 5% and 1% levels.

Table 4 presents the Pearson correlation coefficients between the variables. Both measures of CPG (PHI_D and PHI_M) are significantly and positively correlated with the cost of equity capital (R_{AVE}) . This result suggests that a firm's cost of equity capital increases with the possibility and the level of CPG engagement. It is noteworthy that R_{AVE} is positively correlated with SOE, which means that state-owned enterprises have higher costs of equity capital. In addition, consistent with the prior studies (Xu et al., 2015; Dhaliwal et al., 2016), we find that RAVE is positively correlated with the book-to-market ratio (BTM), leverage (LEV), and long-term growth rate (LTG), whereas it is negatively correlated with a firm's market value of equity (MVE), momentum (MTUM), and liquidity (LIQUID).

4. EMPIRICAL RESULTS

4.1. Univariate tests

Table 5 shows the results of the univariate tests. We partition the sample by $PHI_{-}D$ and analyze the data using mean comparison t-tests and Wilcoxon rank-sum test. The cost of equity capital $(R_{AVE}, R_{GLS}, R_{CT}, R_{PEG}, \text{ and } R_{OJN})$ is lower for firms with $PHI_{-}D = 0$, which provides evidence that CPG is associated with a higher cost of equity capital. These results are consistent with hypothesis 1a. Advertising intensity (ADVERT) is higher when firms engage in CPG activities, suggesting that firms that spend more on advertising also tend to give more to charity (McWilliams and Siegel, 2000; Zhang et al., 2010). In addition, Table 5 demonstrates that MVE, BTM, and LEV are higher for firms that make donations, whereas the other control variables (MTUM, LIQUID, and LTG) are higher for firms that do not make donations.

TABLE 5.Univariate tests.

		0		
	Independent s	amples t-test	Wilcoxon ra	ink-sum test
	$PHI_{-}D = 0 \text{ VS}$	S. $PHI_{-}D = 1$	$PHI_{-}D = 0 \text{ V}$	S. $PHI_{-}D = 1$
	t-statistic	Significance	Z statistic	Significance
R_{AVE}	-6.9070	0.0000	-7.0944	0.0000
R_{GLS}	-8.6216	0.0000	-8.6729	0.0000
R_{CT}	-6.2576	0.0000	-6.9016	0.0000
R_{PEG}	-6.3676	0.0000	-6.4304	0.0000
R_{OJN}	-6.3235	0.0000	-6.3792	0.0000
ADVERT	-5.2570	0.0000	-3.4189	0.0006
MVE	-8.3214	0.0000	-8.2347	0.0000
BTM	-7.3561	0.0000	-7.2374	0.0000
LEV	-6.8753	0.0000	-6.8128	0.0000
MTUM	3.3449	0.0008	3.4671	0.0005
LIQUID	7.6870	0.0000	7.3454	0.0000
LTG	5.4967	0.0000	4.8337	0.0000

4.2. Multivariate regressions

4.2.1. The effects of CPG on the cost of equity capital

To test hypothesis 1, we run the following regression model:

$$R_{i,t} = \beta_0 + \beta_1 PHI_D_{i,t}(PHI_M_{i,t}) + \beta_2 MVE_{i,t} + \beta_3 BTM_{i,t} + \beta_4 LEV_{i,t}$$

$$+ \beta_5 MTUM_{i,t} + \beta_6 LIQUID_{i,t} + \beta_8 LTG_{i,t} + \gamma YearFE_t + \delta IndustryFE_i + \xi G_t$$

where $R_{i,t}$ is the cost of equity capital measures. Industry dummies (IndustryFE) and year dummies (YearFE) are included in the model

to control for unobservable macro-economic factors and industry characteristics. We use the firm fixed-effects method to address the model misspecification problem, and standard errors are clustered at the firm level.

 $\begin{tabular}{ll} \bf TABLE~6. \end{tabular} \label{table energy}$ The effect of CPG on the cost of equity capital.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	R_A	VE	R_G	LS	R_0	CT	R_P	EG	R_C	JN
PHI_D	0.0025^{***}		0.0011^{***}		0.0032^{***}		0.0027^{***}		0.0026^{***}	
	(3.8991)		(3.7587)		(3.8963)		(3.4072)		(3.3812)	
PHI_M		0.0003^{***}		0.0001^{***}		0.0004^{***}		0.0003^{***}		0.0003***
		(5.9825)		(5.4560)		(5.7903)		(5.5022)		(5.4733)
MVE	-0.0001	-0.0003	-0.0022***	-0.0023^{***}	-0.0002	-0.0005	0.0012	0.0009	0.0012	0.0009
	(-0.0787)	(-0.4025)	(-5.9196)	(-6.1852)	(-0.2635)	(-0.5812)	(1.4881)	(1.1767)	(1.4386)	(1.1277)
BTM	0.0164^{***}	0.0162^{***}	0.0226^{***}	0.0225^{***}	0.0158^{***}	0.0155^{***}	0.0141^{***}	0.0139^{***}	0.0138^{***}	0.0135^{***}
	(8.3809)	(8.2940)	(20.3448)	(20.2973)	(6.2717)	(6.1874)	(5.9470)	(5.8521)	(5.7741)	(5.6787)
LEV	0.0110^{***}	0.0109^{***}	0.0060^{***}	0.0059^{***}	0.0109^{***}	0.0107^{***}	0.0139^{***}	0.0138^{***}	0.0139^{***}	0.0138^{***}
	(5.0342)	(4.9863)	(4.8117)	(4.7732)	(3.8304)	(3.7839)	(5.2204)	(5.1737)	(5.2130)	(5.1665)
MTUM	-0.0074^{***}	-0.0074***	-0.0042^{***}	-0.0042^{***}	-0.0094***	-0.0094***	-0.0081^{***}	-0.0081^{***}	-0.0080***	-0.0080***
	(-15.6940)	(-15.6599)	(-16.8137)	(-16.7803)	(-15.5844)	(-15.5552)	(-13.8294)	(-13.7963)	(-13.6414)	(-13.6085)
LIQUID	-0.0002	-0.0002	-0.0001	-0.0001	-0.0001	-0.0001	-0.0004*	-0.0004**	-0.0004**	-0.0004**
	(-1.5349)	(-1.5855)	(-0.7304)	(-0.7717)	(-0.4728)	(-0.5178)	(-1.9434)	(-1.9921)	(-2.0986)	(-2.1476)
LTG	0.0390^{***}	0.0391^{***}	-0.0013^{***}	-0.0013^{***}	0.0491^{***}	0.0492^{***}	0.0504^{***}	0.0505^{***}	0.0514^{***}	0.0514^{***}
	(45.1495)	(45.2488)	(-3.2498)	(-3.1864)	(43.3649)	(43.4870)	(45.4614)	(45.5509)	(46.0853)	(46.1752)
$_CONS$	0.0567^{***}	0.0591^{***}	0.0538^{***}	0.0548^{***}	0.0612^{***}	0.0642^{***}	0.0500^{***}	0.0527^{***}	0.0549^{***}	0.0575^{***}
	(4.9225)	(5.1548)	(8.4648)	(8.6429)	(4.2715)	(4.4878)	(3.2541)	(3.4527)	(3.5589)	(3.7586)
IndustryFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\overline{N}	12612	12612	12612	12612	12612	12612	12612	12612	12612	12612
$R^2(Within)$	0.5297	0.5308	0.5962	0.5969	0.5007	0.5017	0.4900	0.4911	0.4983	0.4993

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

The results are reported in Table 6. The dependent variable in columns (1) and (2) is R_{AVE} which is the average of the four cost of equity capital estimates, and the dependent variables in columns (3)-(10) are R_{GLS} , R_{CT} , R_{PEG} , and R_{OJN} , respectively. The measure for CPG in columns (1), (3), (5), (7), and (9) is PHI_D , while the measure for CPG in columns (2), (4), (6), (8), and (10) is PHI_M . The results show that the coefficients of the CPG measures in all columns are positively and statistically significant at the 1% level, suggesting that firms suffer an increase in the cost of equity capital when they engage in CPG activities or spend more on CPG

activities. These results support hypothesis 1a and are consistent with Judd and Lusch (2018). We also find that the cost of equity capital is negatively associated with the market value of equity, momentum, and liquidity; and positively associated with book-to-market ratio, leverage, and long-term growth rate, similar to the results of the previous studies (Xu et al., 2015; Dhaliwal et al., 2016). However, the coefficients on MVE and LIQUID lose their statistical significance in some regressions.

 ${\it 4.2.2.} \quad {\it The~effects~of~public~awareness~on~the~CPG~-~cost~of~equity~capital~relation}$

We test hypothesis 2 by estimating the following model:

$$R_{i,t} = \beta_0 + \beta_1 PHI_-D_{i,t}(PHI_-M_{i,t}) + \beta_2 PHI_-D_{i,t}(PHI_-M_{i,t}) \times HighAwareness_{i,t}$$

$$+ \beta_3 HighAwareness_{i,t} + \beta_4 MVE_{i,t} + \beta_5 BTM_{i,t} + \beta_6 LEV_{i,t} + \beta_7 MTUM_{i,t}$$

$$+ \beta_8 LIQUID_{i,t} + \beta_9 LTG_{i,t}$$

$$+ \gamma Y earFE_t + \delta IndustryFE_i + \varepsilon_{i,t}$$

$$(7)$$

The equation is estimated based on the firm fixed-effects model and standard errors are clustered at the firm level. The regression results are reported in Table 7. The dependent variable in columns (1) and (2) is R_{AVE} , which is the average of the four cost of equity capital estimates. The dependent variables in columns (3)-(10), are R_{GLS} , R_{CT} , R_{PEG} , and R_{OJN} , respectively. The measure for CPG in columns (1), (3), (5), (7) and (9) is PHI_D , and the measure for CPG in columns (2), (4), (6), (8) and (10) is PHI₋M. The results in columns (1) and (2) are the most important because the mean of the four different estimates of the cost of equity capital can mitigate measurement errors. There are two main facts from columns (1) and (2) that deserve to be highlighted. First, the coefficients on the CPG proxies are still significantly positive, which means that CPG leads to a higher cost of equity capital. Second, the coefficients on the interaction terms of HighAwareness and CPG proxies are significantly negative, which suggests that the positive effect of CPG on a firm's cost of equity capital is alleviated for firms with a high level of public awareness, thereby supporting hypothesis 2a. In addition, we find that the coefficients on HighAwareness are significantly positive. Servaes and Tamayo (2013) show that firms' advertising expenses negatively affect their operating return on sales after controlling for firm fixed effects. We argue that overspending on advertising raises a firm's sales expenses and reduces its profitability, and hence increases a firm's business risk.

 ${\bf TABLE~7.}$ The effect of public awareness on the CPG — cost of equity capital relation.

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)
	R_A	VE	R_G	LS	R_0	CT	R_P	EG	R_O	JN
$\overline{PHI_{-}D}$	0.0029***		0.0012***		0.0038***		0.0030***		0.0029***	
	(4.1370)		(3.6604)		(4.1682)		(3.4305)		(3.3870)	
PHI_M		0.0003^{***}		0.0001^{***}		0.0004^{***}		0.0004^{***}		0.0004***
		(6.0024)		(5.2975)		(5.7316)		(5.3652)		(5.3149)
HighAwareness	0.0040**	0.0038**	0.0006	0.0007	0.0059^{***}	0.0049**	0.0039**	0.0041**	0.0038**	0.0039**
	(2.5096)	(2.3920)	(0.7383)	(0.7801)	(3.0447)	(2.4998)	(2.0181)	(2.1261)	(1.9613)	(2.0725)
$PHI_{-}D*$	-0.0029^*		-0.0005		-0.0044**		-0.0023		-0.0021	
HighAwareness	s(-1.8657)		(-0.5584)		(-2.2892)		(-1.1978)		(-1.1424)	
PHI_M*		-0.0002^*		-0.0000		-0.0003^*		-0.0002		-0.0002
HighAwareness	3	(-1.7129)		(-0.6212)		(-1.6727)		(-1.3046)		(-1.2504)
MVE	-0.0000	-0.0003	-0.0022^{***}	-0.0023***	-0.0002	-0.0005	0.0012	0.0010	0.0012	0.0009
	(-0.0626)	(-0.3839)	(-5.8962)	(-6.1610)	(-0.2451)	(-0.5581)	(1.5075)	(1.1971)	(1.4582)	(1.1483)
BTM	0.0166^{***}	0.0164^{***}	0.0226^{***}	0.0225^{***}	0.0160^{***}	0.0157^{***}	0.0143^{***}	0.0141^{***}	0.0140^{***}	0.0137^{***}
	(8.4092)	(8.3194)	(20.2824)	(20.2288)	(6.3270)	(6.2411)	(5.9946)	(5.8963)	(5.8230)	(5.7242)
LEV	0.0111^{***}	0.0109^{***}	0.0060^{***}	0.0059^{***}	0.0109^{***}	0.0107^{***}	0.0139^{***}	0.0138^{***}	0.0139^{***}	0.0138^{***}
	(5.0440)	(4.9842)	(4.8134)	(4.7719)	(3.8409)	(3.7816)	(5.2264)	(5.1715)	(5.2187)	(5.1643)
MTUM	-0.0074***	-0.0074***	-0.0042^{***}	-0.0042^{***}	-0.0094***	-0.0094***	-0.0081^{***}	-0.0080***	-0.0080***	-0.0079^{***}
	(-15.6401)	(-15.6137)	(-16.8110)	(-16.7783)	(-15.5294)	(-15.5118)	(-13.7778)	(-13.7466)	(-13.5915)	(-13.5605)
LIQUID	-0.0002	-0.0002	-0.0001	-0.0001	-0.0001	-0.0001	-0.0003^*	-0.0004**	-0.0004**	-0.0004**
	(-1.5139)	(-1.5689)	(-0.7231)	(-0.7663)	(-0.4496)	(-0.4985)	(-1.9197)	(-1.9717)	(-2.0752)	(-2.1273)
LTG	0.0391^{***}	0.0391^{***}	-0.0013^{***}	-0.0013***	0.0492^{***}	0.0493^{***}	0.0505^{***}	0.0505^{***}	0.0514^{***}	0.0515^{***}
	(45.2106)	(45.3085)	(-3.2361)	(-3.1658)	(43.3888)	(43.5319)	(45.5080)	(45.5891)	(46.1293)	(46.2109)
$_CONS$	0.0561^{***}	0.0584^{***}	0.0536^{***}	0.0547^{***}	0.0602^{***}	0.0634^{***}	0.0494^{***}	0.0520^{***}	0.0542^{***}	0.0568***
	(4.8725)	(5.1016)	(8.4398)	(8.6209)	(4.2319)	(4.4560)	(3.1965)	(3.3886)	(3.5006)	(3.6939)
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\overline{N}	12612	12612	12612	12612	12612	12612	12612	12612	12612	12612
$R^2(Within)$	0.5300	0.5311	0.5962	0.5969	0.5012	0.5021	0.4904	0.4915	0.4986	0.4996

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

4.2.3. SOEs vs. NSOEs

In the context of the Chinese capital market, one of the most significant features is the existence of SOEs. Because the government remains in control of pivotal resources and persistently intervenes with the operations of SOEs, SOEs and NSOEs differ in terms of their characteristics, strategies, and performance (Liu, 2006; Zhang et al., 2009; Li et al., 2015). SOEs

cannot concentrate on profit maximization as they have to fulfill political and social objectives such as full employment and social stability (Zhang et al., 2009; Chen et al., 2011b; Xu et al., 2015). Hence, the motives and economic consequences of CPG may also vary between SOEs and NSOEs. Thus, we divide the sample into SOEs and NSOEs. The results of the univariate test are reported in Table 8.

TABLE 8.Comparison tests: SOEs vs. NSOEs.

	Mean (SOEs)	Mean (NSOEs)	Difference	Median (SOEs)	Median (NSOEs)	Difference
	N = 5064	N = 7548		N = 5064	N = 7548	
$\overline{R_{AVE}}$	0.0890	0.0852	0.0037***	0.0852	0.0825	0.0027***
PHI_M	11.5273	11.3778	0.1495^{**}	12.6492	12.4244	0.2248^{***}
ADVERT	0.0020	0.0028	-0.0008**	0.0000	0.0002	-0.0002^{***}
MVE	15.8979	15.6534	0.2445^{***}	15.8470	15.6257	0.2213^{***}
BTM	0.6729	0.5694	0.1035^{***}	0.6942	0.5719	0.1223^{***}
LEV	0.5057	0.3688	0.1369^{***}	0.5246	0.3522	0.1724^{***}
MTUM	0.2629	0.2078	0.0551^{***}	0.0611	0.0000	0.0611^{***}
LIQUID	3.0809	3.1160	-0.0351	2.4949	2.6228	-0.1279^{***}
LTG	0.3914	0.4279	-0.0365^{***}	0.2706	0.3366	-0.0660^{***}

Notes: *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

Table 8 shows that both the mean and median of RAVE of SOEs are higher than those of NSOEs. Because of ambiguous property rights, less effective corporate governance mechanisms, and the existence of non-economic objectives in SOEs, they are more likely to suffer higher levels of business risk and cost of equity capital (Qiang, 2003; Zhang et al., 2009; Xu et al., 2015). Table 8 also demonstrates that both the mean and median of PHI_-M for the SOE group are significantly higher, which indicates that SOEs tend to make larger donations than NSOEs. On average, SOEs have a higher market value of equity (MVE), book-to-market ratio (BTM), leverage (LEV), and momentum (MTUM). Meanwhile, they have lower advertising intensity (ADVERT), liquidity (LIQUID), and long-term growth rate (LTG).

We present the regression results of model (6) and (7) in Table 9. The results in columns (1)-(4) show that the effect of CPG on the cost of equity capital is positive and significant for both SOEs and NSOEs, which is consistent with hypothesis 1a. In columns (5)-(8), all the coefficients on the interaction terms ($PHI_D*HighAwareness$) have negative signs, but only the coefficients in columns (7) and (8) are significant. These results indicate that the role of public awareness in mit-

 ${\bf TABLE~9.}$ The effect of public awareness on the CPG — cost of equity capital relation: SOEs vs. NSOEs.

			SOES	vs. Nooles.				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	SC	Es	NSC	OEs		θEs	NSO	OEs
PHI_D	0.0033***		0.0020***		0.0035***		0.0025***	
	(3.1080)		(2.6386)		(2.9290)		(3.0643)	
PHI_M		0.0003^{***}		0.0003^{***}		0.0003^{***}		0.0003^{***}
		(3.6952)		(4.8919)		(3.3465)		(5.2343)
HighAwareness					0.0058**	0.0047^{*}	0.0022	0.0027
					(2.3551)	(1.8482)	(1.1117)	(1.3898)
PHI_D*					-0.0017		-0.0033^*	
HighAwareness					(-0.6907)		(-1.6873)	
PHI_M*						-0.0000		-0.0003**
HighAwareness						(-0.2193)		(-2.0325)
MVE	-0.0004	-0.0006	0.0001	-0.0001	-0.0004	-0.0005	0.0001	-0.0001
	(-0.3522)	(-0.5272)	(0.1344)	(-0.0990)	(-0.3336)	(-0.5017)	(0.1097)	(-0.1195)
BTM	0.0182***	0.0181***	0.0174^{***}	0.0171***	0.0186***	0.0185***	0.0173***	0.0169***
	(5.8372)	(5.8148)	(6.6286)	(6.5436)	(5.9313)	(5.9069)	(6.5613)	(6.4705)
LEV	0.0023	0.0022	0.0167^{***}	0.0165^{***}	0.0024	0.0023	0.0168***	0.0166^{***}
	(0.5747)	(0.5425)	(6.1520)	(6.0905)	(0.6061)	(0.5678)	(6.1835)	(6.1224)
MTUM	-0.0060***	-0.0060****	-0.0077^{***}	-0.0077^{***}	-0.0059^{***}	-0.0059***	-0.0077^{***}	-0.0077^{***}
	(-7.5488)	(-7.5279)	(-12.8680)	(-12.8484)	(-7.4538)	(-7.4415)	(-12.8662)	(-12.8477)
LIQUID	-0.0000	-0.0000	-0.0005^{***}	-0.0005^{***}	-0.0000	-0.0000	-0.0005^{***}	-0.0005^{***}
	(-0.1155)	(-0.1403)	(-2.5985)	(-2.6534)	(-0.0370)	(-0.0531)	(-2.5807)	(-2.6404)
LTG	0.0380***	0.0381***	0.0399***	0.0399***	0.0380***	0.0381***	0.0399***	0.0399***
	(29.0107)	(29.1400)	(34.0080)	(34.0156)	(29.1421)	(29.2667)	(33.9849)	(33.9954)
$_CONS$	0.0659^{***}	0.0685^{***}	0.0496***	0.0517^{***}	0.0643***	0.0670***	0.0489***	0.0508***
	(3.5105)	(3.6630)	(3.4850)	(3.6302)	(3.4186)	(3.5772)	(3.4374)	(3.5679)
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\overline{N}	5064	5064	7548	7548	5064	5064	7548	7548
$R^2(Within)$	0.4879	0.4885	0.5760	0.5772	0.4896	0.4902	0.5762	0.5776

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

igating the effect of CPG on the cost of equity capital is not effective for SOEs. The CPG decisions in SOEs are usually made based on political and social objectives rather than shareholder welfare maximization (Long et al., 2020). In contrast to NSOEs, SOEs are less likely to use CPG as a strategic mechanism to attain legitimacy or build political connections for their long-term development (Li et al., 2015). Therefore, SOEs have a

higher probability of misappropriating money on CPG and damaging the interests of investors compared with NSOEs (Chen et al., 2020). Although advertising raises stakeholder's awareness of a firm's CPG involvement, investors of SOEs might not regard CPG as a favorable signal, and thus the perceived risk of SOEs will not decrease.

 $\begin{tabular}{ll} \textbf{TABLE 10.} \\ \textbf{The effect of public awareness on the CPG---cost of equity capital relation:} \\ \textbf{Eastern region vs. non-eastern region.} \\ \end{tabular}$

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	Eastern	region	Non-easte	rn region	Eastern	region	Non-easte	rn region
PHI_D	0.0028***		0.0016		0.0033***		0.0016	
	(3.9331)		(1.2080)		(4.3703)		(1.0744)	
PHI_M		0.0003^{***}		0.0002^{**}		0.0004^{***}		0.0002^{*}
		(5.8493)		(2.0794)		(6.0608)		(1.8886)
HighAwareness					0.0041**	0.0035^{*}	0.0041	0.0045
					(2.1770)	(1.9347)	(1.3571)	(1.4512)
PHI_D*					-0.0042**		-0.0005	
HighAwareness					(-2.2483)		(-0.1657)	
PHI_M*						-0.0003^*		-0.0001
HighAwareness						(-1.9531)		(-0.3121)
MVE	-0.0007	-0.0009	0.0016	0.0014	-0.0007	-0.0009	0.0017	0.0015
	(-0.8986)	(-1.1521)	(1.3474)	(1.1785)	(-0.9173)	(-1.1642)	(1.4409)	(1.2740)
BTM	0.0160***	0.0158***	0.0148***	0.0146***	0.0160***	0.0158***	0.0154***	0.0151***
	(6.7076)	(6.6472)	(4.2148)	(4.1645)	(6.7025)	(6.6407)	(4.3110)	(4.2550)
LEV	0.0124^{***}	0.0123***	0.0102^{**}	0.0101**	0.0124^{***}	0.0123***	0.0105^{**}	0.0104**
	(4.8511)	(4.8015)	(2.4144)	(2.3953)	(4.8547)	(4.7872)	(2.4603)	(2.4437)
MTUM	-0.0077^{***}	-0.0077^{***}	-0.0070^{***}	-0.0070^{***}	-0.0077^{***}	-0.0077^{***}	-0.0070^{***}	-0.0070^{***}
	(-14.3361)	(-14.3093)	(-7.3153)	(-7.2943)	(-14.3009)	(-14.2792)	(-7.2971)	(-7.2817)
LIQUID	-0.0003	-0.0003	-0.0002	-0.0002	-0.0003	-0.0003	-0.0002	-0.0002
	(-1.5386)	(-1.5805)	(-0.6181)	(-0.6397)	(-1.5124)	(-1.5506)	(-0.6502)	(-0.6780)
LTG	0.0393^{***}	0.0393^{***}	0.0385^{***}	0.0386^{***}	0.0393^{***}	0.0393^{***}	0.0386^{***}	0.0387^{***}
	(36.0374)	(36.1113)	(27.0575)	(27.1076)	(36.0458)	(36.1134)	(27.1253)	(27.1733)
$_CONS$	0.0716^{***}	0.0735***	0.0324	0.0347^{*}	0.0711^{***}	0.0730***	0.0304	0.0326
	(3.7919)	(3.9254)	(1.6143)	(1.7335)	(3.7756)	(3.9029)	(1.5126)	(1.6270)
Industry FE	Yes							
YearFE	Yes							
\overline{N}	8754	8754	3858	3858	8754	8754	3858	3858
$R^2(Within)$	0.5531	0.5544	0.4919	0.4925	0.5534	0.5547	0.4933	0.4938

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

4.2.4. Eastern region vs. non-eastern region

Although China's economy has made great progress over the past forty years, substantial imbalances still exist among different regions. Compared with the central and western regions, the eastern region is relatively well-developed in terms of its institutional environment and has a higher level of marketization (Wang and Qian, 2011). The non-eastern region with weak institutional environments is confronted with more severe government resource control and poorer enforcement of laws and regulations (Huang and Rice, 2012; Chen et al., 2018b). These institutional deficiencies in the non-eastern region can be exploited by opportunistic firms for their own benefits. Hence, the motives of CPG are more complicated for firms located in the non-eastern region. We divide our sample into eastern region and non-eastern region groups based on the region where the firm's headquarter is located. We then rerun the regressions based on models (6) and (7).

The results are reported in Table 10. For the eastern region group, the coefficients of the CPG proxies remain significantly positive, and the coefficients of interaction terms remain significantly negative, which are consistent with our main findings. However, for the non-eastern region, the coefficients of PHI_D and the interaction terms are not significant. The coefficient on PHI_M for the non-eastern region group is significantly positive, but its magnitude and t-statistics are smaller than those of the PHI_M for the eastern region group. Because of the poor institutional environment and high government intervention, firms located in the non-eastern region are more likely to engage in CPG activities to trade for some essential resources or hide misbehavior (Chen et al., 2018b). Hence, firms located in the non-eastern region suffer less business risk and enjoy more economic benefits when they donate, such that the positive effect of CPG on the cost of equity capital is not as obvious as with firms in the eastern region.

4.2.5. High market competition vs. low market competition

Market competition is considered as an important factor that influences a firm's CPG initiatives. However, the effects of market competition are still controversial. Zhang et al. (2010) find that firms in highly competitive industries are more likely to use CPG to differentiate themselves from their rivals. On the contrary, Bagnoli and Watts (2003) suggest that high market competition prevents firms from investing in CSR because engaging in CSR increases the marginal cost of producing private goods. Hence, it

is necessary to examine the effects of CPG and public awareness on the firm's cost of equity capital under different market competition conditions.

 $\begin{tabular}{ll} \textbf{TABLE 11.} \\ \textbf{The effect of public awareness on the CPG---cost of equity capital relation:} \\ \textbf{High market competition vs. low market competition.} \\ \end{tabular}$

	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
	High market	competition	Low market	competition	High market	competition	Low market	competition
$PHI_{-}D$	0.0031***		0.0009		0.0032***		0.0019	
	(3.9422)		(0.8736)		(3.7580)		(1.6220)	
$PHI_{-}M$		0.0004^{***}		0.0002**		0.0004^{***}		0.0002^{***}
		(5.5895)		(2.0535)		(5.3259)		(2.6039)
High Awareness					0.0030	0.0035^{*}	0.0066**	0.0055**
					(1.5432)	(1.8258)	(2.4374)	(2.0820)
PHI_D*					-0.0011		-0.0064**	
High Awareness					(-0.5781)		(-2.4113)	
PHI_M*						-0.0001		-0.0004**
High Awareness						(-0.8910)		(-1.9995)
MVE	0.0004	0.0002	-0.0010	-0.0012	0.0004	0.0002	-0.0010	-0.0012
	(0.4536)	(0.2110)	(-0.9756)	(-1.1311)	(0.5044)	(0.2642)	(-0.9747)	(-1.1526)
BTM	0.0162^{***}	0.0159^{***}	0.0176^{***}	0.0175^{***}	0.0165^{***}	0.0162^{***}	0.0175^{***}	0.0174^{***}
	(6.7333)	(6.6516)	(4.9091)	(4.8860)	(6.8007)	(6.7198)	(4.8706)	(4.8446)
LEV	0.0076^{***}	0.0074^{***}	0.0176^{***}	0.0176^{***}	0.0076^{***}	0.0073^{***}	0.0177^{***}	0.0177^{***}
	(2.8694)	(2.7884)	(4.2201)	(4.2199)	(2.8633)	(2.7744)	(4.2362)	(4.2411)
MTUM	-0.0072^{***}	-0.0072***	-0.0075***	-0.0075^{***}	-0.0072^{***}	-0.0072^{***}	-0.0075^{***}	-0.0075^{***}
	(-12.3513)	(-12.3569)	(-9.1447)	(-9.1093)	(-12.3191)	(-12.3200)	(-9.1073)	(-9.0766)
LIQUID	-0.0001	-0.0001	-0.0004*	-0.0004^*	-0.0001	-0.0001	-0.0004^*	-0.0004*
	(-0.5678)	(-0.5867)	(-1.6620)	(-1.7101)	(-0.5219)	(-0.5456)	(-1.6622)	(-1.7125)
LTG	0.0387^{***}	0.0388^{***}	0.0393^{***}	0.0394^{***}	0.0388^{***}	0.0388^{***}	0.0393^{***}	0.0394^{***}
	(36.4932)	(36.5819)	(26.9173)	(26.9638)	(36.5783)	(36.6677)	(26.9016)	(26.9322)
$_CONS$	0.0509^{***}	0.0534^{***}	0.0739^{***}	0.0755***	0.0499^{***}	0.0523^{***}	0.0730^{***}	0.0751^{***}
	(3.7493)	(3.9458)	(3.9208)	(4.0306)	(3.6536)	(3.8430)	(3.8741)	(4.0126)
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
\overline{N}	8236	8236	4376	4376	8236	8236	4376	4376
$R^2(Within)$	0.5462	0.5475	0.5137	0.5143	0.5466	0.5479	0.5145	0.5149

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

We use the Herfindahl-Hirschman Index (HHI) to measure the level of market competition as follows:

$$HHI_{j,t} = \sum_{i=1}^{N_j} S_{i,j,t}^2 \tag{8}$$

where $S_{i,j,t}$ is the market share of firm i in industry j in year t, and the industry classification is based on the codes provided by CSRC. A high HHI represents low market competition, and vice versa. An industry is regarded as having strong market competition if its HHI is under the sample median for a specific fiscal year. Following the prior studies (e.g., Zhang et al., 2010; Chen et al., 2018a; Khan et al., 2020), we calculate HHI based on the sales revenue of all listed firms for each industry. The sales revenue data used to construct HHI is provided by the CSMAR database and covers 34,466 firm-year observations.

Table 11 provides the regression results for the two groups. In columns (1)-(4), the coefficients on both CPG proxies in the high-market-competition group are significantly positive. In the low-market-competition group, the coefficient on PHI₋M remains significantly positive, whereas the coefficient on PHI_D is not significant. These results indicate that the positive effect of CPG on the cost of equity capital is more pronounced in highly competitive industries. Furthermore, in columns (5)-(8), we find that the coefficients on the interaction terms of CPG and public awareness are significantly negative in the low-market-competition group, but not significant in the high-market-competition group. This means that the moderating effect of public awareness only exists in low-competition industries. In high-competition industries, firms' CPG involvement may result in higher "opportunity costs" and business risk because CPG would lower firms' free cash flow and make firms miss good investment opportunities, which are more likely to disappear quickly in high-competition industries (Chen et al., 2018b). Hence, firms in high-competition industries suffer a higher cost of equity capital when engaging in CPG activities compared with those in low-competition industries. In addition, the advertising activities of rivals of homogenous products are more intense in high-competition industries, such that the attention of stakeholders is distracted. As a result, the moderating effect of public awareness is less obvious in high-competition group.

5. ROBUSTNESS TESTS

Our main analysis suggests that CPG positively affects a firm's cost of equity capital, and the effect is mitigated by public awareness. However, the results might suffer from endogeneity problems caused by measurement error or omitted variables that can affect both CPG and the cost of equity capital. We conduct the following robustness tests to confirm the reliability of our results.

 ${\bf TABLE~12.}$ Robustness tests: High advertising intensity vs. low advertising intensity.

1 COD GENTLESS (cous. Ingn acre	reising meensies	vs. low advert	ising intensity.
	(1)	(2)	(3)	(4)
	High advertis	ing intensity	Low advertisi	ing intensity
$\overline{PHI_D}$	-0.0002		0.0034***	
	(-0.1419)		(3.1557)	
PHI_M		0.0001		0.0004***
		(0.7008)		(4.5546)
MVE	-0.0000	-0.0001	-0.0015	-0.0018^*
	(-0.0307)	(-0.0668)	(-1.5914)	(-1.8831)
BTM	0.0073^{*}	0.0074^{*}	0.0165^{***}	0.0162^{***}
	(1.7249)	(1.7450)	(5.5681)	(5.4814)
LEV	0.0041	0.0040	0.0107^{***}	0.0106^{***}
	(0.9119)	(0.8824)	(2.8894)	(2.8568)
MTUM	-0.0130^{***}	-0.0130^{***}	-0.0052^{***}	-0.0052^{***}
	(-11.4937)	(-11.5194)	(-7.6314)	(-7.5999)
LIQUID	-0.0009^{***}	-0.0009^{***}	-0.0004*	-0.0004*
	(-3.0570)	(-3.0531)	(-1.8418)	(-1.9122)
LTG	0.0379^{***}	0.0379^{***}	0.0404^{***}	0.0405^{***}
	(15.8609)	(15.8373)	(32.6031)	(32.7339)
$_CONS$	0.0367	0.0363	0.0899^{***}	0.0934^{***}
	(1.4541)	(1.4456)	(5.3125)	(5.5188)
Industry FE	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes
\overline{N}	2565	2565	6052	6052
$R^2(Within)$	0.5756	0.5757	0.5481	0.5493

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

5.1. High advertising intensity vs. low advertising intensity

To further examine the mitigating effects of public awareness on the CPG's effect on the cost of equity capital, we divide our sample of firms

into four quartiles based on the advertising intensity measure ADVERT and rerun our regressions on the top and bottom quartile groups. Firms in the top (bottom) quartile have a high (low) level of advertising intensity. Table 12 reports the estimates of model (6). The coefficients on PHI_D and PHI_M for the low-advertising-intensity group are both significantly positive at the 1% level, while the coefficients on these two CPG measures are not significant for the high-advertising-intensity group. These results imply that firms with low public awareness suffer a higher cost of equity capital when they engage in CPG activities, which is consistent with our aforementioned arguments.

 $\begin{tabular}{ll} \textbf{TABLE 13.} \\ \textbf{Robustness tests: Large firms vs. small firms.} \\ \end{tabular}$

		0			
	(1)	(2)	(3)	(4)	
	Large firms		Small firms		
PHI_D	0.0016		0.0023**		
	(0.7984)		(2.1896)		
PHI_M		0.0002		0.0002^{***}	
		(1.6180)		(2.7896)	
MVE	0.0006	0.0004	-0.0063^{***}	-0.0064^{***}	
	(0.4186)	(0.2856)	(-3.9446)	(-3.9633)	
BTM	0.0153^{***}	0.0152^{***}	0.0160^{***}	0.0161^{***}	
	(3.0501)	(3.0331)	(3.5456)	(3.5768)	
LEV	-0.0002	-0.0006	0.0216^{***}	0.0217^{***}	
	(-0.0380)	(-0.0930)	(3.7936)	(3.7997)	
MTUM	-0.0092^{***}	-0.0091^{***}	-0.0042^{***}	-0.0042^{***}	
	(-8.1564)	(-8.1328)	(-4.0604)	(-4.0767)	
LIQUID	0.0001	0.0001	-0.0007**	-0.0007**	
	(0.3751)	(0.3420)	(-2.4861)	(-2.5223)	
LTG	0.0355^{***}	0.0356^{***}	0.0439^{***}	0.0439^{***}	
	(19.8889)	(19.9615)	(26.1861)	(26.1598)	
$_CONS$	0.0485^{*}	0.0507^{*}	0.1613^{***}	0.1612^{***}	
	(1.8393)	(1.9255)	(6.2673)	(6.2826)	
Industry FE	Yes	Yes	Yes	Yes	
YearFE	Yes	Yes	Yes	Yes	
\overline{N}	3075	3075	3237	3237	
$R^2(Within)$	0.4795	0.4801	0.6237	0.6244	

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

5.2. Large firms vs. small firms

The prior research suggests that large firms have better visibility, which can attract greater public attention (Seifert et al., 2004; Amato and Amato, 2007; Baldini et al., 2018). Therefore, we divide our sample into four quartiles based on the firms' assets and rerun our regressions on the top and bottom quartile groups. Firms in the top (bottom) quartile have a large (small) size. The regression results of model (6) reported in Table 13 show that CPG can significantly increase the cost of equity capital of small firms, but has no effects on the cost of equity capital of large firms. These results imply that small firms with a low degree of visibility (public awareness) suffer a higher cost of equity capital when they engage in CPG activities. Hence, public awareness can mitigate the positive effect of CPG on the cost of equity capital.

 ${\bf TABLE~14.}$ Robustness tests: An alternative measure of public awareness.

	(1)	(2)	(3)	(4)
$\overline{PHI_{-}D}$	0.0025***		0.0024***	
	(3.8991)		(3.8831)	
PHI_M		0.0003***		0.0003***
		(5.9825)		(5.9312)
Public Awareness			0.0617^{**}	0.0583**
			(2.2984)	(2.1773)
$PHI_D*PublicAwareness$			-0.0000****	
			(-2.8504)	
$PHI_M * PublicAwareness$,	-0.0000***
				(-2.9082)
MVE	-0.0001	-0.0003	-0.0001	-0.0003
	(-0.0787)	(-0.4025)	(-0.1221)	(-0.4385)
BTM	0.0164***	0.0162***	0.0167***	0.0165***
	(8.3809)	(8.2940)	(8.4829)	(8.3920)
LEV	0.0110***	0.0109***	0.0108***	0.0107***
	(5.0342)	(4.9863)	(4.9427)	(4.9011)
MTUM	-0.0074^{***}	-0.0074***	-0.0074***	-0.0073***
	(-15.6940)	(-15.6599)	(-15.6113)	(-15.5808)
LIQUID	-0.0002	-0.0002	-0.0002	-0.0002
	(-1.5349)	(-1.5855)	(-1.5326)	(-1.5829)
LTG	0.0390***	0.0391***	0.0390***	0.0391***
	(45.1495)	(45.2488)	(45.1867)	(45.2824)
$_{-}CONS$	0.0567^{***}	0.0591***	0.0570***	0.0593***
	(4.9225)	(5.1548)	(4.9443)	(5.1730)
IndustryFE	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes
\overline{N}	12612	12612	12612	12612
$R^2(Within)$	0.5297	0.5308	0.5300	0.5310

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

5.3. An alternative measure of public awareness

In this section, we use a more size-free measurement of public awareness to check the robustness of the results. Specifically, we define PublicAwareness as the ratio of advertising expenditures to lagged total revenue and set PublicAwareness equal to the industry median if missing. As shown in Table 14, the coefficients on the interaction terms of CPG and public awareness are still significantly negative, while the absolute values of the inter-

action terms become smaller. These findings indicate that our results do not depend on the specific measurement of public awareness.

5.4. Control for operating performance

The control variables used in our main regressions are mostly measurements of firms' financial performance. However, the level of business risk and the cost of equity capital are also related to a firm's operating performance. To alleviate the concerns of omitted variable bias, we add the return on assets (ROA) and total factor productivity (TFP) into our regressions to control for firms' operating performance.

ROA is calculated as the earnings before interest and tax divided by total assets. TFP represents the transformation efficiency of total inputs into total outputs. To estimate TFP, we refer to prior literature (e.g., Faleye et al., 2006; Dai et al., 2017) and assume that the firm's production function follows the Cobb-Douglas form:

$$Y_{i,t} = AL_{i,t}^{\alpha} K_{i,t}^{\beta} \tag{9}$$

where $Y_{i,t}$ is the sales of firm i in year t; $L_{i,t}$ is the number of employees of firm i in year t; and $K_{i,t}$ is the net property, plant, and equipment of firm i in year t. We implement the natural logarithm of both sides of equation (9), and then calculate the TFP of the firm as the residual using the following regression equation:

$$y_{i,t} = a_{i,t} + \alpha l_{i,t} + \beta k_{i,t} + \varepsilon_{i,t} \tag{10}$$

where $y_{i,t}$, $l_{i,t}$, and $k_{i,t}$ are the logarithmic forms of $Y_{i,t}$, $L_{i,t}$, and $K_{i,t}$, respectively. We estimate equation (10) by industry and year, and exclude regressions with less than 20 observations.

We re-estimate our main regressions after including both ROA and TFP. The results reported in Table 15 show that our main regression results remain robust.

 ${\bf TABLE~15.}$ Robustness tests: Controlling for operating performance.

	(1)	(2)	(3)	(4)
PHI_D	0.0020***		0.0024***	
	(3.1566)		(3.4448)	
PHI_M		0.0003***		0.0003^{***}
		(4.9992)		(5.1512)
HighAwareness			0.0042**	0.0042^{**}
			(2.5514)	(2.5727)
$PHI_D*HighAwareness$			-0.0029^*	
			(-1.7899)	
$PHI_M*HighAwareness$				-0.0002^*
				(-1.8021)
MVE	-0.0025^{***}	-0.0027^{***}	-0.0025^{***}	-0.0027^{***}
	(-3.5279)	(-3.7776)	(-3.4761)	(-3.7232)
BTM	0.0245^{***}	0.0243^{***}	0.0247^{***}	0.0244^{***}
	(11.4757)	(11.3792)	(11.5261)	(11.4244)
LEV	0.0173^{***}	0.0171^{***}	0.0172^{***}	0.0171^{***}
	(7.1275)	(7.0741)	(7.1172)	(7.0553)
MTUM	-0.0082^{***}	-0.0082^{***}	-0.0082^{***}	-0.0081^{***}
	(-17.6699)	(-17.6185)	(-17.6021)	(-17.5620)
LIQUID	0.0001	0.0001	0.0001	0.0001
	(0.4136)	(0.3723)	(0.4336)	(0.3899)
LTG	0.0442^{***}	0.0442^{***}	0.0442^{***}	0.0442^{***}
	(45.1564)	(45.2296)	(45.2279)	(45.2972)
TFP	0.0017^{**}	0.0017^{**}	0.0018**	0.0018**
	(2.4514)	(2.4373)	(2.4939)	(2.4778)
ROA	0.1310^{***}	0.1304^{***}	0.1306^{***}	0.1300^{***}
	(9.5294)	(9.4923)	(9.5099)	(9.4718)
$_CONS$	0.0698^{***}	0.0718^{***}	0.0690^{***}	0.0710^{***}
	(4.5145)	(4.6655)	(4.4650)	(4.6114)
Industry FE	Yes	Yes	Yes	Yes
YearFE	Yes	Yes	Yes	Yes
\overline{N}	11584	11584	11584	11584
$R^2(Within)$	0.5695	0.5704	0.5699	0.5708

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

5.5. Propensity score matching

We implement a propensity score matching method to address the endogeneity concerns arising from selection bias. First, we divide our sam-

TABLE 16.

Robustness tests: Propensity score matching.

	(1)	(2)
$\overline{PHI_BIGGIVER}$	0.0027***	0.0033***
	(3.7419)	(4.1550)
HighAwareness		0.0027^{*}
		(1.7715)
$PHI_BIGGIVER*HighAwareness$		-0.0033**
		(-2.1068)
MVE	0.0004	0.0005
	(0.4070)	(0.4418)
BTM	0.0220^{***}	0.0220^{***}
	(6.9852)	(7.0047)
LEV	0.0105^{***}	0.0102^{***}
	(2.9831)	(2.9109)
MTUM	-0.0061***	-0.0060***
	(-7.2502)	(-7.2279)
LIQUID	-0.0005**	-0.0005**
	(-2.0319)	(-2.0583)
LTG	0.0390^{***}	0.0390^{***}
	(26.8018)	(26.8063)
$_{-}CONS$	0.0644^{***}	0.0632^{***}
	(3.6889)	(3.6345)
Industry FE	Yes	Yes
YearFE	Yes	Yes
\overline{N}	5086	5086
$R^2(Within)$	0.5501	0.5507

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

ple into two subsamples based on the sample mean of the firms' donation expenditures. We construct a new CPG measure $PHI_BIGGIVER$ that equals 1 if the amount of a firm's donation expenditures is above the sample mean in a specific fiscal year and 0 otherwise. Second, we run a probit regression on our full sample to predict the possibility of donating a larger amount than the sample mean. All the control variables in model (6) are included in the prediction model. We then match each $PHI_BIGGIVER = 1$ observation with a $PHI_BIGGIVER = 0$ observation that has the closest propensity score. Finally, we examine the effects of CPG and public awareness on the cost of equity capital using this

matched sample. The results presented in Table 16 are consistent with our main findings.

TABLE 17.
Robustness tests: 2SLS method.

	Estimates of model (6)		Estimates of model (7)	
	(1)	(2)	(3)	(4)
	First stage	Second stage	First stage	Second stage
$\overline{IndustryAveragePHI_M}$	0.8635***		0.8860***	
	(20.8264)		(20.1255)	
$Industry Average PHI_M*High Awareness$			-0.1731	
			(-1.6410)	
PHI_M		0.0011***		0.0014***
		(4.7251)		(5.7120)
HighAwareness			2.1410^*	0.0426***
			(1.7141)	(4.1518)
$PHI_M*HighAwareness$				-0.0033***
-				(-3.7892)
MVE	0.6065***	-0.0074^{***}	0.5979***	-0.0074***
	(7.7379)	(-14.1810)	(7.6164)	(-14.0101)
BTM	1.0398***	0.0232***	1.0482***	0.0234***
	(3.0707)	(11.4484)	(3.0960)	(11.2973)
LEV	0.2410	0.0115***	0.2383	0.0113***
	(0.5684)	(4.8529)	(0.5621)	(4.7283)
MTUM	-0.0750	-0.0088***	-0.0600	-0.0086^{***}
	(-1.0716)	(-23.0240)	(-0.8514)	(-21.9444)
LIQUID	-0.0165	-0.0016^{***}	-0.0181	-0.0017^{***}
	(-0.6875)	(-11.9864)	(-0.7520)	(-11.9823)
LTG	-0.2433^*	0.0381***	-0.2377^*	0.0383***
	(-1.7427)	(41.3476)	(-1.7035)	(41.1382)
The p-value of the instrument's				
partial F test	0.0000		0.0000	
Underidentification test:				
Kleibergen-Paap rk LM statistic	235.74***		33.39***	
Weak identification test:				
Cragg-Donald Wald F statistic	516.38***		123.77***	
N	12113	12113	12113	12113

Notes: The equations are estimated based on the fixed-effects model and standard errors are clustered at the firm level (t-statistics are in parentheses). *, ** and *** represent significance at the 10%, 5% and 1% levels, respectively.

5.6. Two-stage least squares regression

Because it is possible that CPG and the cost of equity capital are simultaneously affected by some omitted variables, we perform a two-stage least squares (2SLS) regression to deal with the omitted variable concerns. Following Xu et al. (2015) and Judd and Lusch (2018), we use the CPG behavior of industry peers as our instrumental variable measure. Specifically, we define $IndustryAveragePHI_M$ as the industry average of PHI_M in a specific fiscal year. In the first stage, we estimate CPG using the instrument variable and the other control variables included in our main regressions. In the second stage, we use the predicted CPG measure as the independent variable to estimate models (6) and (7). The procedure is performed using a Stata external command xtivreg2, and 499 observations are deleted automatically due to the singleton group problem. The results are presented in Table 17, which are consistent with those in the main tests. Following Xie and Zhang (2020), we conduct several tests to check the validity of the instrument variable. The p-values of the instrument's partial F-test for the first stage are equal to 0.0000, which indicates that the instrument variable is highly correlated with the endogenous variable. In addition, the instrument also passes both the Kleibergen-Paap and the Cragg-Donald relevance tests. Thus, our results are robust after controlling for endogeneity using the 2SLS method.

6. CONCLUSION

This study examines the influence of CPG on the cost of equity capital and investigates whether public awareness affects the effect of CPG on the cost of equity capital. Using a sample of Chinese listed companies, this study shows that CPG positively affects the cost of equity capital. We also find that this positive effect is mitigated for firms with high public awareness. Our findings are consistent with the prior work suggesting that firms are unable to benefit from CSR involvement without public awareness (Servaes and Tamayo, 2013). However, the moderating effect of public awareness only applies to NSOEs when the sample is divided into SOEs and NSOEs. Furthermore, we find that for firms in the eastern region or those in less competitive industries, the positive effect of corporate philanthropy on the cost of equity capital can be moderated as the firms' public awareness increases.

Our findings provide evidence that CPG activities can increase firm's equity costs and harm stockholders' value in China. Hence, strengthen-

ing corporate governance mechanisms would be beneficial to firms in terms of alleviating agency concerns and costs arising from CPG activities, and enabling firms to protect the interest of shareholders when fulfilling their social responsibilities. Moreover, this study suggests that high public awareness through advertising could help spread good news about firms' CPG activities and mitigate the passive impact of CPG on firms' equity cost, as this can help firms build beneficial and long-term relationships with stakeholders. Furthermore, the findings of this study suggest that the Chinese government's control over market resources may distort the motivations for CPG at the cost of the firms' value and increase the probability of market failure as firms may trade CPG for resources that are necessary for their development. Lastly, this study benefits investors as it serves as a reminder that a firm's CPG activities should be taken into consideration when making investment decisions. Investors need to pay attention to the trade-offs between the costs of CPG and its potential benefits.

REFERENCES

Allen, Franklin, Jun Qian, and Meijun Qian, 2005. Law, finance, and economic growth in China. *Journal of Financial Economics* **77**, 57-116.

Amato, Louis H. and Christie H. Amato, 2007. The effects of firm size and industry on corporate giving. *Journal of Business Ethics* **72**, 229-241.

Amihud, Yakov and Haim Mendelson, 1986. Asset pricing and the bid-ask spread. *Journal of Financial Economics* 17, 223-249.

Aramburu, Izaskun A. and Irune G. Pescador, 2019. The effects of corporate social responsibility on customer loyalty: The mediating effect of reputation in cooperative banks versus commercial banks in the Basque Country. *Journal of Business Ethics* **154**, 701-719.

Ashbaugh-Skaife, Hollis, Daniel W. Collins, William R. Kinney Jr, and Ryan Lafond, 2009. The effect of SOX internal control deficiencies on firm risk and cost of equity. *Journal of Accounting Research* 47, 1-43.

Atkinson, Lisa and Joseph Galaskiewicz, 1988. Stock ownership and company contributions to charity. Administrative Science Quarterly 33, 82-100.

Bagnoli, Mark and Susan G. Watts, 2003. Selling to socially responsible consumers: Competition and the private provision of public goods. *Journal of Economics and Management Strategy* 12, 419-445.

Balakrishnan, Ramji, Geoffrey B. Sprinkle, and Michael G. Williamson, 2011. Contracting benefits of corporate giving: An experimental investigation. *The Accounting Review* 86, 1887-1907.

Baldini, Maria, Lorenzo D. Maso, Giovanni Liberatore, Francesco Mazzi, and Simone Terzani, 2018. Role of country- and firm-level determinants in environmental, social, and governance disclosure. *Journal of Business Ethics* **150**, 79-98.

Berman, Shawn L., Andrew C. Wicks, Suresh Kotha, and Thomas M. Jones, 1999. Does stakeholder orientation matter? The relationship between stakeholder management models and firm financial performance. *Academy of Management Journal* 42, 488-506.

Boubakri, Narjess, Omrane Guedhami, Dev Mishra, and Walid Saffar, 2012. Political connections and the cost of equity capital. *Journal of Corporate Finance* 18, 541-559.

Boubakri, Narjess, Omrane Guedhami, and Walid Saffar, 2016. Geographic location, foreign ownership, and cost of equity capital: Evidence from privatization. *Journal of Corporate Finance* 38, 363-381.

Brammer, Stephen and Andrew Millington, 2005. Corporate reputation and philanthropy: An empirical analysis. *Journal of Business Ethics* **61**, 29-44.

Brogaard, Jonathan, Dan Li, and Ying Xia, 2017. Stock liquidity and default risk. *Journal of Financial Economics* **124**, 486-502.

Brown, William O., Eric Helland, and Janet K. Smith, 2006. Corporate philanthropic practices. *Journal of Corporate Finance* **12**, 855-877.

Chen, Donghua, Dequan Jiang, and Xin Yu, 2015. Corporate philanthropy and bank loans in China. *Pacific-Basin Finance Journal* **35**, 402-424.

Chen, Jun, Wang Dong, Jamie Tong, and Feida Zhang, 2018a. Corporate philanthropy and tunneling: Evidence from China. *Journal of Business Ethics* **150**, 135-157.

Chen, Jun, Wang Dong, Jamie Y. Tong, and Feida F. Zhang, 2018b. Corporate philanthropy and investment efficiency: Empirical evidence from China. *Pacific-Basin Finance Journal* **51**, 392-409.

Chen, Jun, Wang Dong, Yixing Tong, and Feida Zhang, 2020. Corporate philanthropy and corporate misconduct: Evidence from China. *International Review of Economics and Finance* **65**, 17-31.

Chen, Kevin C. W., Zhihong Chen, and John Wei, 2011a. Agency costs of free cash flow and the effect of shareholder rights on the implied cost of equity capital. *Journal of Financial and Quantitative Analysis* 46, 171-207.

Chen, Shimin, Zheng Sun, Song Tang, and Donghui Wu, 2011b. Government intervention and investment efficiency: Evidence from China. *Journal of Corporate Finance* 17, 259-271.

Chu, Teresa, In-Mu Haw, Simon S. M. Ho, and Xu Zhang, 2019. Labor protection, ownership concentration, and cost of equity capital: International evidence. *Review of Quantitative Finance and Accounting* **54**, 1351-1387.

Chu, Teresa, In-Mu Haw, Bryan B. Lee, and Woody Wu, 2014. Cost of equity capital, control divergence, and institutions: The international evidence. *Review of Quantitative Finance and Accounting* **43**, 483-527.

Claus, James and Jacob Thomas, 2001. Equity premia as low as three percent? Evidence from analysts' earnings forecasts for domestic and international stock markets. *Journal of Finance* **56**, 1629-1666.

Coval, Joshua D. and Tobias J. Moskowitz, 1999. Home bias at home: Local equity preference in domestic portfolios. *Journal of Finance* **54**, 2045-2073.

Dai, Yunhao, Dongmin Kong, and Jin Xu, 2017. Does fairness breed efficiency? Pay gap and firm productivity in China. *International Review of Economics and Finance* 48, 406-422.

Dhaliwal, Dan S., Oliver Z. Li, Albert Tsang, and Yong G. Yang, 2011. Voluntary nonfinancial disclosure and the cost of equity capital: The initiation of corporate social responsibility reporting. *The Accounting Review* **86**, 59-100.

Dhaliwal, Dan, J. Scott Judd, Matthew Serfling, and Sarah Shaikh, 2016. Customer concentration risk and the cost of equity capital. *Journal of Accounting and Economics* **61**, 23-48.

Du, Xingqiang, 2015. Is corporate philanthropy used as environmental misconduct dressing? Evidence from Chinese family-owned firms. *Journal of Business Ethics* **129**, 341-361.

Easton, Peter D., 2004. PE ratios, PEG ratios, and estimating the implied expected rate of return on equity capital. *The Accounting Review* **79**, 73-95.

Esch, Franz-Rudolf, Tobias Langner, Bernd H. Schmitt, and Patrick Geus, 2006. Are brands forever? How brand knowledge and relationships affect current and future purchases. *Journal of Product and Brand Management* 15, 98-105.

Faleye, Olubunmi, Vikas Mehrotra, and Randall Morck, 2006. When labor has a voice in corporate governance. *Journal of Financial and Quantitative Analysis* **41**, 489-510.

Falkinger, Josef, 2008. Limited attention as a scarce resource in information-rich economies. *The Economic Journal* **118**, 1596-1620.

Fama, Eugene F. and Kenneth R. French, 1992. The cross-section of expected stock returns. *Journal of Finance* 47, 427-465.

Fee, C. Edward, Charles J. Hadlock, and Joshua R. Pierce, 2009. Investment, financing constraints, and internal capital markets: Evidence from the advertising expenditures of multinational Firms. *Review of Financial Studies* **22**, 2361-2392.

Finkel, Eli J., Michael I. Norton, Harry T. Reis, Dan Ariely, Peter A. Caprariello, Paul W. Eastwick, Jeana H. Frost, and Michael R. Maniaci, 2015. When does familiarity promote versus undermine interpersonal attraction? A proposed integrative model from erstwhile adversaries. *Perspectives on Psychological Science* 10, 3-19.

Fiske, Susan T. and Martha G. Cox, 1979. Person concepts: The effect of target familiarity and descriptive purpose on the process of describing others. *Journal of Personality* 47, 136-161.

Foroudi, Pantea, Zhongqi Jin, Suraksha Gupta, Mohammad M. Foroudi, and Philip J. Kitchen, 2018. Perceptional components of brand equity: Configuring the symmetrical and asymmetrical paths to brand loyalty and brand purchase intention. *Journal of Business Research* 89, 462-474.

French, Kenneth R. and James M. Poterba, 1991. Investor diversification and international equity markets. *American Economic Review* 81, 222-226.

Gao, Fox, Robert Faff, and Farshid Navissi, 2012. Corporate philanthropy: Insights from the 2008 Wenchuan earthquake in China. *Pacific-Basin Finance Journal* **20**, 363-377.

Gao, Xinghua and Yonghong Jia, 2017. The role of internal control in the equity issue market: Evidence from seasoned equity offerings. *Journal of Accounting, Auditing and Finance* **32**, 303-328.

Gardberg, Naomi A., Stelios C. Zyglidopoulos, Pavlos C. Symeou, and Donald H. Schepers, 2019. The impact of corporate philanthropy on reputation for corporate social performance. *Business and Society* 58, 1177-1208.

Gebhardt, William R., Charles M. C. Lee, and Bhaskaran Swaminathan, 2001. Toward an implied cost of capital. *Journal of Accounting Research* **39**, 135-176.

Godfrey, Paul C., 2005. The relationship between corporate philanthropy and shareholder wealth: A risk management perspective. *Academy of Management Review* **30**, 777-798.

Grullon, Gustavo, George Kanatas, and James P. Weston, 2004. Advertising, breadth of ownership, and liquidity. *Review of Financial Studies* 17, 439-461.

Guay, Wayne, S. P. Kothari, and Susan Shu, 2011. Properties of implied cost of capital using analysts' forecasts. *Australian Journal of Management* 36, 125-149.

Gupta, Atul, Kartik Raman, and Chenguang Shang, 2018. Social capital and the cost of equity. *Journal of Banking and Finance* 87, 102-117.

Ha, Hong Y. and Helen Perks, 2005. Effects of consumer perceptions of brand experience on the web: Brand familiarity, satisfaction and brand trust. *Journal of Consumer Behaviour* 4, 438-452.

Hail, Luzi and Christian Leuz, 2006. International differences in the cost of equity capital: Do legal institutions and securities regulation matter? *Journal of Accounting Research* 44, 485-531.

Hogarth, Kate, Marion Hutchinson, and Wendy Scaife, 2018. Corporate philanthropy, reputation risk management and shareholder value: A study of Australian corporate giving. *Journal of Business Ethics* **151**, 375-390.

Huang, Fang and John Rice, 2012. Firm networking and bribery in China: Assessing some potential negative consequences of firm openness. *Journal of Business Ethics* **107**, 533-545.

Jones, Edward E. and Keith E. Davis, 1965. From acts to dispositions the attribution process in person perception. Advances in Experimental Social Psychology 2, 219-266.

Judd, J. Scott and Stephen J. Lusch, 2018. Philanthropy and the cost of equity capital. Working Paper, University of Illinois at Chicago. Available at SSRN: https://ssrn.com/abstract=3040170.

Kashmiri, Saim, Prachi Gala, and Cameron D. Nicol, 2019. Seeking pleasure or avoiding pain: Influence of CEO regulatory focus on firms' advertising, R&D, and marketing controversies. *Journal of Business Research* **105**, 227-242.

Kato, Takao and Cheryl Long, 2006. CEO turnover, firm performance, and enterprise reform in China: Evidence from micro data. *Journal of Comparative Economics* **34**, 796-817.

Keller, Kevin L., 1993. Conceptualizing, measuring, and managing customer-based brand equity. *Journal of Marketing* 57, 1-22.

Khan, Muhammad A., Xuezhi Qin, Khalil Jebran, and Irfan Ullah, 2020. Uncertainty and R&D investment: Does product market competition matter? Research in International Business and Finance 52. https://doi.org/10.1016/j.ribaf.2019.101167.

Khlif, Hichem, Khaled Samaha, and Mark Soliman, 2019. Internal control quality, voluntary disclosure, and cost of equity capital: The case of an unregulated market. *International Journal of Auditing* **23**, 144-160.

Kim, Jeong-Bon, Mary L. Z. Ma, and Haiping Wang, 2015. Financial development and the cost of equity capital: Evidence from China. *China Journal of Accounting Research* 8, 243-277.

Laroche, Michel, Chankon Kim, and Lianxi Zhou, 1996. Brand familiarity and confidence as determinants of purchase intention: An empirical test in a multiple brand context. *Journal of Business Research* 37, 115-120.

Lee, Ki-Hoon and Dongyoung Shin, 2010. Consumers' responses to CSR activities: The linkage between increased awareness and purchase intention. *Public Relations Review* $\bf 36$, 193-195.

Lev, Baruch, Christine Petrovits, and Suresh Radhakrishnan, 2010. Is doing good good for you? How corporate charitable contributions enhance revenue growth. *Strate-gic Management Journal* **31**, 182-200.

Levy, Haim and Moshe Levy, 2014. The home bias is here to stay. $Journal\ of\ Banking\ and\ Finance\ {f 47},\ 29\text{-}40.$

Li, Sihai, Xianzhong Song, and Huiying Wu, 2015. Political connection, ownership structure, and corporate philanthropy in China: A strategic-political perspective. *Journal of Business Ethics* **129**, 399-411.

Li, Shaofang and Chao Liu, 2018. Quality of corporate social responsibility disclosure and cost of equity capital: Lessons from China. *Emerging Markets Finance and Trade* **54**, 2472-2494.

Liket, Kellie and Karen Maas, 2016. Strategic philanthropy: Corporate measurement of philanthropic impacts as a requirement for a "happy marriage" of business and society. *Business and Society* **55**, 889-921.

Liu, Qiao, 2006. Corporate governance in China: Current practices, economic effects and institutional determinants. CESifo Economic Studies 52, 415-453.

Long, Wenbin, Sihai Li, Huiying Wu, and Xianzhong Song, 2020. Corporate social responsibility and financial performance: The roles of government intervention and market competition. *Corporate Social Responsibility and Environmental Management* 27, 525-541.

Maas, Karen and Kellie Liket, 2011. Talk the walk: Measuring the impact of strategic philanthropy. *Journal of Business Ethics* **100**, 445-464.

Macdonald, Emma K. and Byron M. Sharp, 2000. Brand awareness effects on consumer decision making for a common, repeat purchase product: A replication. *Journal of Business Research* 48, 5-15.

Mandhachitara, Rujirutana and Yaowalak Poolthong, 2011. A model of customer loyalty and corporate social responsibility. *Journal of Services Marketing* **25**, 122-133

Marquis, Christopher, Mary A. Glynn, and Gerald F. Davis, 2007. Community isomorphism and corporate social action. Academy of Management Review 32, 925-945.

Marquis, Christopher and Matthew Lee, 2013. Who is governing whom? Executives, governance, and the structure of generosity in large U.S. firms. *Strategic Management Journal* 34, 483-497.

Masulis, Ronald W. and Syed W. Reza, 2015. Agency problems of corporate philanthropy. Review of Financial Studies 28, 592-636.

McWilliams, Abagail and Donald Siegel, 2000. Corporate social responsibility and financial performance: Correlation or misspecification? *Strategic Management Journal* **21**, 603-609.

McWilliams, Abagail and Donald Siegel, 2001. Corporate social responsibility: A theory of the firm perspective. Academy of Management Review 26, 117-127.

Meng, Xiangtong, Wei Zhang, Youwei Li, Xing Cao, and Xu Feng, 2020. Social media effect, investor recognition and the cross-section of stock returns. *International Review of Financial Analysis* **67**. https://doi.org/10.1016/j.irfa.2019.101432.

Mizuno, Mitsuru, 2010. Institutional investors, corporate governance and firm performance in Japan. *Pacific Economic Review* **15**, 653-665.

Moser, Donald V. and Patrick R. Martin, 2012. A broader perspective on corporate social responsibility research in accounting. *The Accounting review* 87, 797-806.

Nieuwerburgh, Stijn V. and Laura Veldkamp, 2009. Information immobility and the home bias puzzle. *Journal of Finance* **64**, 1187-1215.

Ohlson, James A., 1995. Earnings, book values, and dividends in equity valuation. Contemporary Accounting Research 11, 661-687.

Ohlson, James A. and Beate E. Juettner-Nauroth, 2000. Expected EPS and EPS growth as determinants of value. Working paper, New York University.

Ohlson, James A. and Beate E. Juettner-Nauroth., 2005. Expected EPS and EPS growth as determinants of value. *Review of Accounting Studies* **10**, 349-365.

Petrenko, Oleg V., Federico Aime, Jason Ridge, and Aaron Hill, 2016. Corporate social responsibility or CEO narcissism? CSR motivations and organizational performance. *Strategic Management Journal* 37, 262-279.

Pham, Anh V., 2019. Political risk and cost of equity: The mediating role of political connections. *Journal of Corporate Finance* **56**, 64-87.

Pomering, Alan and Sara Dolnicar, 2009. Assessing the prerequisite of successful CSR implementation: Are consumers aware of CSR initiatives? *Journal of Business Ethics* **85**, 285-301.

Qiang, Qu, 2003. Corporate governance and state-owned shares in China listed companies. *Journal of Asian Economics* 14, 771-783.

Reis, Harry T., Michael R. Maniaci, Peter A. Caprariello, Paul W. Eastwick, and Eli J. Finkel, 2011. Familiarity does indeed promote attraction in live interaction. *Journal of personality and social psychology* **101**, 557-570.

Rhou, Yinyoung, Manisha Singal, and Yoon Koh, 2016. CSR and financial performance: The role of CSR awareness in the restaurant industry. *International Journal of Hospitality Management* 57, 30-39.

Richardson, Alan J. and Michael Welker, 2001. Social disclosure, financial disclosure and the cost of equity capital. *Accounting, Organizations and Society* 26, 597-616.

Riff, Sivan and Yossi Yagil, 2020. Home bias and the power of branding. *Journal of Behavioral Finance* **22**, 1-9.

Saiia, David H., Archie B. Carroll, and Ann K. Buchholtz, 2003. Philanthropy as strategy: When corporate charity "begins at home". *Business and Society* **42**, 169-201.

Sánchez, Carol M., 2000. Motives for corporate philanthropy in El Salvador: Altruism and political legitimacy. *Journal of Business Ethics* **27**, 363-375.

Scherer, Andreas G. and Guido Palazzo, 2007. Toward a political conception of corporate responsibility: Business and society seen from a Habermasian perspective. *Academy of Management Review* **32**, 1096-1120.

Schrader, Anke and Minji Xie, 2016. Corporate philanthropy in China: Creating impact amid challenges. *The Conference Board China Center*. Available at: https://www.conference-board.org/topics/china-UN-SDGs.

Schuler, Douglas A. and Margaret Cording, 2006. A corporate social performance-corporate financial performance behavioral model for consumers. *Academy of Management Review* 31, 540-558.

Seifert, Bruce, Sara A. Morris, and Barbara R. Bartkus, 2004. Having, giving, and getting: Slack resources, corporate philanthropy, and firm financial performance. *Business and Society* **43**, 135-161.

Servaes, Henri and Ane Tamayo, 2013. The impact of corporate social responsibility on firm value: The role of customer awareness. *Management Science* **59**, 1045-1061.

Stulz, René M., 1999. Golbalization, corporate finance, and the cost of capital. *Journal of Applied Corporate Finance* 12, 8-25.

Varadarajan, P. Rajan and Anil Menon, 1988. Cause-related marketing: A coalignment of marketing strategy and corporate philanthropy. *Journal of marketing* **52**, 58-74.

Wang, Heli, Jaepil Choi, and Jiatao Li, 2008. Too little or too much? Untangling the relationship between corporate philanthropy and firm financial performance. *Organization Science* 19, 143-159.

Wang, Heli and Cuili Qian, 2011. Corporate philanthropy and corporate financial performance: The roles of stakeholder response and political access. *Academy of Management Journal* **54**, 1159-1181.

Wang, Jia and Betty S. Coffey, 1992. Board composition and corporate philanthropy. *Journal of Business Ethics* 11, 771-778.

Wang, Shuo, Yuhui Gao, Gerard P. Hodgkinson, Denise M. Rousseau, and Patrick C. Flood, 2015. Opening the black box of CSR decision making: A policy-capturing study of charitable donation decisions in China. *Journal of Business Ethics* **128**, 665-683.

Werbel, James D. and Suzanne M. Carter, 2002. The CEO's influence on corporate foundation giving. *Journal of Business Ethics* $\bf 40$, 47-60.

Xie, Jun, 2015. CEO career concerns and investment efficiency: Evidence from China. *Emerging Markets Review* **24**, 149-159.

Xie, Jun and Yifan Zhang, 2020. Anti-corruption, government intervention, and corporate cash holdings: Evidence from China. *Economic Systems* 44. https://doi.org/10.1016/j.ecosys.2020.100745.

Xu, Shan, Duchi Liu, and Jianbai Huang, 2015. Corporate social responsibility, the cost of equity capital and ownership structure: An analysis of Chinese listed firms. *Australian Journal of Management* **40**, 245-276.

Yang, Yuan and Min Tang, 2020. Finding the ethics of "red capitalists": Political connection and philanthropy of Chinese private entrepreneurs. *Journal of Business Ethics* **161**, 133-147.

Yung, Kenneth and Nadia Nafar, 2017. Investor attention and the expected returns of reits. *International Review of Economics and Finance* 48, 423-439.

Zhang, Min, Lu Xie, and Haoran Xu, 2016. Corporate philanthropy and stock price crash risk: Evidence from China. *Journal of Business Ethics* **139**, 595-617.

Zhang, Ran, Zabihollah Rezaee, and Jigao Zhu, 2009. Corporate philanthropic disaster response and ownership type: Evidence from Chinese firms' response to the Sichuan earthquake. *Journal of Business Ethics* **91**, 51-63.

Zhang, Ran, Jigao Zhu, Heng Yue, and Chunyan Zhu, 2010. Corporate philanthropic giving, advertising intensity, and industry competition level. *Journal of Business Ethics* **94**, 39-52.

Zolotoy, Leon, Don O'Sullivan, and Jill Klein, 2019. Character cues and contracting costs: The relationship between philanthropy and the cost of capital. *Journal of Business Ethics* **154**, 497-515.